Return to: KMB Enterprises 17204 Hwy 126 Sisters, OR 97759 STATE OF OREGON) SS.
COUNTY OF DESCRUTES) SS.
1, MARY SUF PENHOLLOW, COUNTY CLERK AN RECORDER OF CONVEYANCES. IN AND FOR SAIL INSTRUMENT WAS RECORDED THIS DAY:

On lan.

93 MAR -- 9 PM 2: 52 MARY SUE PENHOLLOY? COUNTY CLER (

BY. M. SALLE DEPUT NO. 93-0705.4 REAL SOCI

93-07054

292 - 0713

DECLARATION

o£

COVENANTS, CONDITIONS, AND RESTRICTIONS

for

ASPEN LAKES

TABLE OF CONTENTS

Article	I pai	OTET TOTAL	ege
		1.111042	
	1.1	TSPRU TIPE	• • • 1
	1.2	ASPEN LAKES"	
	1.3	Area of Common Responsibility". Articles of Incorporation".	1.,
	1.4	Articles of Incorporation"	1
	1.5	Association	.1
	1.6	Base Assessment" Board of Directors or "Board"	.1
	17	Board of Directors or "Board"	1.
	1.8	"Anildor" "Bosiness" and "Trade"	١.
	1.0	Business and "Trade"	.2
		By Laws "Class B' Control Period".	.2
	1.10	"Class "B" Control Period"	2
	1.11	Common Area*	2
	1.12	"Common Expenses". "Community-Wide Standard".	2
	1.13	"Community-Wide Standard"" "Declarant"	2
	1.14	"Declarant" "Design Guideljnes"	Ž
	1.15	"Design Guidelines"" "Golf Course"	2
	1.16	"Golf Course"	1
	1.17	"Golf Course"	•
	1.18	"daringn Goits"	•
	1.19	"Header"	
	1.20	"Member"	
	1.21	"Mortgage"	
	1.22	Mortgagee"	
	1.23	"Hortgagor"	
	1.24	"Neighborhood"	
	1.25	"Office of the Goanty Clerk"	
	1.26	Open Space"	
	1.27	"O.R.S"	
	1.28	"Owner"	
	1.29	"Persoa"	
	1.30	Private Amenities	
	1.31	Private Amenities"	
	1.32	Properties	
	1.33	Retail Owner"	
	1.34	"Special Assessment"	
	1.35	Specific Assessment 3 Supplemental Declaration 3	
		Supplemental Declaration 3	
	1.36	Unit	
	1.37	"Voting Bember or Hembers"	

article II	LVAL DET!	RIGETS	
	2.1		
	2.2	Cornor Area	
	٤.٤	Private Amerities	
Article III	ASSOCIATI	OF PURCTION, MEMBERSHIP AND VOTING RIGHTS	
	3.1		
		Vanction of Association	

		3.2 Membership.
		3.2 Membership. 3.3 Yoting.
		3.3 Yoting.
Artic	ele IV	BYONE AND
	,	RIGHTS AND OBLIGATIONS OF THE ASSOCIATION
		4.1 Comment
		4.1 Common Area. 4.2 Personal Property and Real Property for Common Use. 4.3 References.
		4.2 Personal Property and Real Property for Common Use. 4.3 References. 4.4 Implied Rights: Roard Lab.
		4.3 References
		4.3 Leforcement. 4.4 Implied Rights; Board Authority. 5.5 Governmental Interest:
		4.4 Implied Rights; Board Authority
		4.5 Governmental Interests
		.7 Dedication of Common Areas
	•	.8 Security
		.6 Security
Article		
uttīcīē	, į	I Association's Bossess
		The state of the s
	5.,	Owner's Responsibility
	5.3	Owner's Responsibility. Standard of Performance.
	5.	Standard of Performance
		Party Walls and Similar Structures
årticle V	I IAS	MARCE ARD CASUALTY LOSSES
		waterstrand the contract of the contract
	6.1	
	6.2	Omners Insurance
	6.3	Madeke and Horrenseas
	6.4	Disbursement of Proceeds
	6.5	Repair and Passarbunat
		Repair and Reconstruction
Article VII	HO PA	TITION
1	_	······································
Article VIII	CONDE	DATION
		BATION11
Article 19		The state of the s
mitters II	AUREXA	TION AND WITEDRAYAL OF PROPERTY
	9.1	
	9.2	NUCCALLED BITHOUT ADDRESS T C
		Approval of Memberchin
	9.3	WILEGERMAL OF Property
	9.4	#WGILIONAL Mayonante and p
	9.5	VARADE DE
	9.6	Amendment
		Fassing of Construction
Article X		
WATELE Y	ASSESSEE	MS
	10.1	
	.417	oreation of Accessore
	10.2	Gapitalization of Association.

	101.	- Part of commendation of treatments
	10.4	Date of Commencement of Assessments
	10.9	Computation of Base Assessment
	10.6	Reserve Budget and Capital Contribution
	10.7	Special Assessments14
	10.8	Limitation of Increases of Assessment States of States o
	10.9	Limitation of Increases of Assessments
	10.1	A Lian for Annual Control of the Con
	10.1	O Lien for Assessments
	2911.	Exempt Property
Artisle Y	I ADMUS	
	· aavmi	PROTIDAL CTANALDRE
	11.1	Manage 1
		General
	11.2	Architectural Review
	11.3	Guidelines and Procedures
	11.4	Submission of Plans and Specifications
	11.5	Bo Haires of B.
	11.6	No Waiver of Puture Approvals
	11.7	Variance
		Limitation of Cizbility
	11.8	Koforcement
		Margin
Article III	use cu	IDELINES AND RESTRICTIONS
	12.1	Plan of Development; Applicability; Bffect
	12.2	Board Power.
	12.3	Board Power
	12.4	Bembers' Power
	12.5	Owners Acknowledgment
	12.5	Rights of Owners
	14.0	Initial Use Guidelines and Restrictions
Article XIII		
werters Will	KASULAT	522
	13.1	French of Bonn b
	13.2	Basements of Encroachment
	13.3	Basements for Willities, Btc
		Easements to Serve Additional Property
	13.4	Essements for Golf Course
	13.5	
	13.6	Right of Botry. .23 Baintenance .23
	13.7	Kaintananea
		Baintenance
tticle XIT	evitgage	PROFIS (085
	14.1	Motices of Action
	14.2	
	14.3	
	14.7	Amendment by Board

	14.1 14.1	· APPLICADILITY of total
	144)	Pailure of Mortgagee to Respond

Article	Ye aper	Mamala
		iriki 'S RIGHTS
	15 .	
	13,1	Transfer
	15.2	Ownership and Operation of Cate of
	15.3	Right To Use
	15.4	Ownership and Operation of Golf Course
	15,5	dimitations on Assault
	15.6	Jurisdiction and Connection
		and boobersting,
		77
Article 17	I disput	RESOLUTION FOR THE PROPERTY.
		AND DIMITATION ON LITIGATION
	16.1	R RESOLUTION AND LIMITATION ON LITICATION
	16.2	Agreement to Avoid Costs of Litigation and to Limit Right to Litigate Disputes
	16.3	Exempt Claims
	16.4	Handatory Procedures For All Other Claims
	16.5	Allocation of Costs of Resolving Claims
	10.7	Allocation of Costs of Resolving Claims
		20
Article XV[]	ABSON	
Transport Hill	OPERAL	PROFIZIORS
		PROFISIONS
	17.2	Amendment
	17.3	
	17.4	
	17.8	Compliance
	17.9	
		Inforcement of Bonded Obligations

TABLE OF EXHIBITS

	Land Initially Submitted.	e Pirst Pearing
	Land Subject to Ammeration. By-Lams of Aspen Lakes Estate Owners, Ioc. Rules of Arbitration.	2

DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS

FOR

ASPEN LAKES

THIS DECLARATION OF COTERANTS, CONDITIONS, AND RESTRICTIONS FOR ASPEN LAKES ("Declaration") is made this & day of March, 1993, by KMB Enterprises, an Oregon partnership ("Declarant").

Declarant filed the Declarations, Restrictions, Protective Covenants and Conditions for COLP COURSE ESTATES AT ASPEN LAKES, recorded in Book #237, Page #1406, in the Official Records in the Office of the County Clerk, Deschutes County, Oregon;

Declarant is still the owner of the real property described in Exhibit "A" and "E" attached to and made part of said Declaration.

Declarant, hereby, amends and replaces in its entirety, all language in aforementioned filing, including the title, with this DECLARATION OF COTEMANTS, CONDITIONS, AND RESTRICTIONS FOR ASPEN LAKES.

Declarant is the owner of the real property described in Biblik "A", which is attached and incorporated by reference. This Declaration imposes upon the Properties (as defined in Article I) mutually beneficial restrictions under a general plan of improvements for the benefit of the owners of each portion of the Properties, and establishes a flexible and reasonable procedure for the overall development, administration, maintenance and preservation of the Properties.

Declarant hereby declares that all of the property described in Exhibit "A" and any additional property subjected to this Declaration by Supplemental Declaration (as defined in Article I) shall be held, sold, used and conveyed subject to the following easements, restrictions, covenants, and conditions, which are for the purpose of protecting the value and desirability of and which shall run with the real property subject to this Declaration. This Declaration shall be binding on and shall inure to the benefit of all parties having any right, title, or interest in the Properties or any part thereof, their heirs, successors, successors-intitle, and assigns.

Article I

The Terms used in this Declaration shall generally be given their natural, commonly accepted definitions except as otherwise specified. Capitalized terms shall be defined as set forth below.

- 1.1 ASPER LAKES": The Properties as described in Section 1.31.
- 1.2 "Area of Common Responsibility": The Common Area, together with those areas, if any, which by the terms of this Declaration, any Supplemental Declaration or other applicable covenants, or by contract become the responsibility of the Association.
- 1.3 "Articles of Incorporation" or "Articles": The Articles of Incorporation of ASPEE LARES ESTATE OWNERS, INC. as filed with the Secretary of State, Corporation Division of the Business Registry for the State of Oregon.
- 1.4 "Association": ASPEN LAKES ESTATE OFFERS, INC. an Oregon comprofit nutual benefit corporation, its successors and assigns.
- 1.5 "Base Assessment": Assessments levied on all Units to fund Common Expenses for the general benefit of all Units, as more particularly described in Sections 10.1, 10.3, and 10.4.
- 1.6 "Board of Directors" or "Board": The body responsible for administration of the Association, selected as provided in the By-Laws and generally serving the same role as the board of directors under the Oregon Momprofit Corporations Act and the Oregon Planned Community Act.

1

- 1.7 "Builder": Any Person which purchases one or more, Units for the purpose of constructing improvements for later sale to consumers, or parcels of land within the Properties for further subdivision, development, and/or resale in the ordinary course of such Ferson's business.
- 1.8 "Business" and "Trade": Shall be construed to have their ordinary, generally accepted meanings, and shall include, without limitation, any occupation, work, or activity undertaken on an ongoing basis which involves the provision of goods or services to Persons other than the provider's family and for which the provider receives a fee, compensation, or other form of consideration, prefit, or (c) a license is required.
- 1.9 By-Laws*: The By-Laws of ASPBM LAKES ESTATE OWNERS, INC. attached as Exhibit "C" and incorporated by reference, as they may be amended.
- 1.10 "Class "B" Control Period": The period of time during which the Class "B" member is entitled to appoint a majority of the Board of Directors as provided in Section 3.1 of the By-Laws.
- 1.11 "Common Area": All real and personal property which the Association owns, leases or otherwise holds possessory or use rights in for the common use and enjoyment of the Owners.
- 1.12 "Common Brpenses": The actual and estimated expenses incurred or acticipated to be incurred by the Association for the general benefit of all Units, including any reasonable reserve, as the Board may find necessary and appropriate pursuant to this Declaration, the By-Laws, and the Articles of Incorporation. Common Expenses shall not include any expenses incurred during the Class "8" Courto! Period for initial development, original construction, installation of infrastructure, original capital improvements, or other original construction costs unless approved by Voting Members representing a majority of the total Class "A" vote of the Association.
- 1.13 "Community Wide Standard": The standard of conduct, maintenance, or other activity generally prevailing throughout the Properties. Such standard may be more specifically determined by the Board of Directors and the Design Review Committee.
- 1.14 "Declarant": KMB EMTERPRISES, an Oregon partnership, or any successor, successor in-title, or assign who has or takes title to any portion of the property described on Mahibits "A" or "B" for the purpose of development and/or resale in the ordinary course of business and who is designated as the Declarant in a recorded instrument executed by the immediately preceding Declarant.
- 1.15 *Design Guidelines*: The architectural guidelines and procedures adopted by the Design Review Committee pursuant to Article XI and applicable to all Units within the Properties.
- 1.16 "Golf Course": Any parcel of land adjacent to or within the Properties which is privately owned by KBB Enterprises, an Oregon partnership, its successors, successors—in—title, or assigns, and which is operated as a golf course, and all related and supporting facilities and improvements operated in connection with such golf course.
- "Master Plan": The Master Plan for the Development of ASPER LAXES as approved by CU-89-79 on the 8th day of August, 1989, as it may be amended, which plan includes the property described on Exhibit "A" and the property described on Exhibit "B" which Declarant may from time to time anticipate subjecting to this Declaration. The Master Plan may also include subsequent plans approved by Deschutes County for the development of all or a portion of the property described on Exhibit "B" which Declarant may from time to time anticipate subjecting to this Declaration. Inclusion of property on the Master Plan shall not, under any circumstances, obligate Declarant to subject such property to this Declaration nor shall the exclusion of property described on Exhibit "B" from the Master Plan bar its later annexation in accordance with Article IX. It is anticipated by the Master Plan that declarant may decide in the future to convert the Properties described in this Section to a "Destination Resort" as that term is defined in the Deschutes County Zoning Ordinance.
- 1.18 "Maximum Units": The units approved for development under the Maxter Plan approved by CO-89-70 on the 8th day of August, 1989, plus the maximum number of Units which may be developed to the property described on Exhibit "B" under the zoning

designation of such property from time to time.

- 1.19 <u>Member</u>: A Person entitled to membership in the Association, as provided in Section 3.2.
- 1.20 **Mortgage*: A mortgage, a deed of trust, a deed to secure debt, or any other form of security deed.
- 1.21 "Mortgagee": A beneficiars or holder of a Mortgage.
- 1,22 "Mortgagor": Any Person who gives a Mortgage.
- 1.23 "Neighborhood": Two or more Units which share interests other than those common to all Units. By way of illustration and not limitation, a condominium, townshome development, cluster home development, or single-family detached housing development might be designated as separate Heighborhoods, or a Neighborhood may be comprised of more than one housing type with other features to division into more than one Neighborhood upon development as any of the above shall constitute a Neighborhood, subject
 - 1.24 Office of the County Clerk": The Office of the County Clerk of Deschutes County, Oregon.
- 1.25 "Open Space": Privately owned land designated as "Open Space" and further regulated as such by Deschutes County Zoning Ordinances. Such designation may be placed on parcels within the Properties of Aspen Lakes.
 - 1.26 O.R.S. The Oregon Revised Statutes.
- 1.27 **Owner**: One or more Persons who hold the record title to any Unit, but excluding in all cases any party holding an interest merely as security for the performance of an obligation. If a Unit is sold under a recorded contract of sale, then upon recording of such contract, the purchaser (rather than the fee owner) will be considered the Owner, if the contract specifically so provides.
 - 1.28 Person: A natural person, a corporation, a partnership, a trustee, or any other legal entity.
- 1.29 "Phase": All Units simultaneously subjected to this Declaration by the Declarant by its execution and recordation of this Declaration and each Supplemental Declaration in the Office of the County Glerk. The property described in Exhibit "A" of this Declaration shall constitute the first Phase ("Phase I").
- 1.30 Private Amenities. Certain real property and any improvements and facilities thereon located adjacent to, in the vicinity of, or within the Properties, which are privately owned and operated by Persons other than the Association for recreational course, if any.

 Course, if any.
- 1.31 "Properties": The real property described in Exhibit "A", together with such additional property as is subject to this Declaration in accordance with Article IX. Exhibit "A" and the Supplemental Declaration which subject additional property to the Declaration shall provide a legal description of the Common Area, if-any.
 - 1.32 Retail Owner: An Owner other than the Declarant or Builder.
 - 1.33 Special Assessment*: Assessments levied in accordance with Section 10.6.
 - 1.34 "Specific Assessment": Assessments levied in accordance with Section 10.7.
- 1.35 "Supplemental Declaration": An amendment or supplement to this Declaration filed pursuant to Article IX which subjects additional property to this Declaration, and identifies the Common Area within the additional property, if any, and/or imposes, expressly or by reference, additional restrictions and obligations on the land described therein.

<u>Dair</u>: A portion of the Properties, whether improved or natuproved, which may be independently owned and conveyed and which is intended for development, use, and occupancy as an attached or detached residence for a single family. The term shall refer to the land, if any, which is part of the Unit as well as any improvements thereon. The term shall include, by way of illustration but not limitation, condominium units, townhouse units, cluster homes, patio or zero lot line homes, and single-family deteched benses on separately platted lots, as well as vacant land intended for development as such, but shall not include Gomeon Areas or property dedicated to the public. In the case of a building within a condominium or other structure containing multiple

In the case of a parcel of vacant land or land on which improvements are under construction, the parcel shall be deemed to contain the number of Units designated for residential use for such parcel on the Master plan or the site plan approved by Declarant, whichever is more recent, until such time as a subdivision plat or condominium plat is filed of record on all or a portion of the parcel. Thereafter, the portion encompassed by such plat shall constitute a separate Unit or Units as determined above and the number of Units on the remaining land, if any shall continue to be determined in accordance with this paragraph.

"Yoting Member or Members": Any Owner, Member, or Declarant representative selected formally or informally as appropriate to represent his/her Unit through signature on petitions, or voice or written ballot, including proxy, according to the

Article II PROPERTY RIGHTS

- 2.1 Common Area. Beery Owner shall have a right and nonexclusive easement of use, access, and enjoyment in and to the Common Area, subject to:
 - (a) This Declaration, the By-Laws and any other applicable covenants;
 - (b) Any restrictions or limitations contained in any deed conveying such property to the Association;
- (c) The right of the Board to adopt rules regulating the use and enjoyment of the Common Area, including rules restricting use of recreational facilities within the Common Area to occupant of Units and their guests and rules limiting the number
- (d) The right of the Board to suspend the right of an Owner to use recreational facilities within the Common Area (i) for any period during which any charge against such Owner's Unit remains delinquent, and (ii) for a period not to exceed 30 days for a single violation or for a longer period in the case of any continuing violation of the Declaration, any applicable Supplemental Declaration, the By-Laws, or rules of the Association after notice and a bearing pursuant to Section 3.23 of the By-Laws;
- (e) The right of the Association, acting through the Board, to dedicate or transfer all or any part of the Common Area pursuant to Section 4.7;
- (f) The right of the Board to impose reasonable membership requirements and charge reasonable membership admission or other fees for the use of any recreational famility situated upon the Common Area;
- (g) The right of the Board to permit use of any recreational facilities situated on the Common Area by persons other than owners, their families, lessees and guests upon payment of use fees established by the Board;
- (h) The right of the Association, acting through the Board, to mortgage, pledge, or hypothecate any or all of its real or personal property as security for money borrowed or debts incorred, subject to the approval requirements set forth in Section

Any Owner may extend his or her right of use and enjoyment to the members of his or her family, permitted lessees, and social invitees, subject to reasonable Board regulation. An Owner who has the right to and does lease his or her Unit shall be deemed to have assigned all such rights to the lessee of such Unit.

The initial Common Area as identified in Exhibit "A" shall be conveyed to the Association prior to or in concurrence with the conveyance of a Unit to a Retail Owner.

2.2 <u>Private Amenities.</u> Access to and use of the Private Amenities is strictly subject to the rules and procedures of the respective Owners of the Private Amenities, and no Person gains any right to enter or to use those facilities by virtue of membership in the Association or ownership or occupancy of a Unit.

All Persons, including all Owners, are hereby advised that no representations or marranties, either written or oral, have been or are made by the Declarant or any other Person with regard to the nature or size of improvements to, or the continuing ownership or operation of the Private Amenities. He purported representation or marrant, written or oral, in conflict with this Section shall be effective without an amendment to this Declaration executed or joined into by the Declarant or the owner(s) of the Private Amenity(ies) which are the subject thereof.

The ownership or operation duties of and as to the Private Amenities may change at any time and from time to time by virtue of, but without limitation, (a) the sale to or assumption of operations by an independent entity, (b) conversion of the numbership structure to an "equity" club or similar arrangement whereby the members of a Private Amenity or an entity owned or controlled thereby become the owner(s) and/or operator(s) of the Private Amenity, or (c) the conveyance of a Private Amenity to one or more affiliates, shareholders, employees, or independent contractors of the Declarant. No consent of the Association or any Owner shall be required to effectuate such a transfer or conversion.

Rights to use the Private Amenities will be granted only to such person, and on such terms and conditions, as may be determined by their respective owners. Such owners shall have the right, from time to time to their sole and absolute discretion and without notice, to amend or waive the terms and conditions of use of their respective Private Amenities and to terminate use rights altogether.

Article III ASSOCIATION FUNCTION, NEMBERSHIP AND VOTING RIGHTS

- 3.1 <u>Function of Association</u>. The Association shall be the entity responsible for management, maintenance, operation and control of the Area of Common Responsibility. The Association shall be the primary entity responsible for enforcement of this Decisration and such reasonable rules regulating use of the Properties as the Board may adopt. The Association shall also be responsible for administering and enforcing the architectural standards and controls set forth in this Decisration and in the Design Guidelines. The Association shall perform its functions in accordance with this Decisration, the By-Laws, the Articles, and Oregon Law.
- 3.2 <u>Membership</u>. Every Owner shall be a Member of the Association. There shall be only one membership per Onit. If a Unit is owned by more than one Person, all co-Owners shall share the privileges of such membership, subject to reasonable Board regulation and the restrictions on voting set forth in Section 3.3 and in the By-Laws, and all such co-Owners shall be jointly and severally obligated to perform the responsibilities of Owners. The membership rights and privileges of an Owner who is a natural person may be exercised by the Member or the Member's spouse. The membership rights of an Owner which is a corporation, partnership or other legal entity may be exercised by any individual designated from time to time by the Owner in a written instrument provided to the Secretary of the Association.
 - 3.3 Yoting. The Association shall have two classes of membership, Class "A" and Class "B".
- (a) Class "A": Class "A" Hembers shall be all Owners except the Class "B" Member, if any. Class "A" Hembers shall have one equal vote for each Unit in which they hold the interest required for membership under Section 3.2; there shall be only one vote per Unit.
- (b) Class "B": Class "8" Members shall be the Declarant and any Builder who are Owners. Class "B" Members shall be entitled to three votes for each Unit owned. Unless otherwise specified in this Declaration, the By-Laws, or an agreement between

Declarant and a Builder, the vote for each Unit owned by Builders shall be exercised by the Declarant. The Class "B" membership shall cease and be converted to Class "A" membership upon the earlier of the following:

(i) When 75% of the Maximum Units are owned by Retail Owners;

(ii) 15 years from the anniversary date of closing of escroy for the first Unit after the first sale to a Retail Owner; provided that the 15-year period for conversion shall be extended an additional five years unless a majority of the Voting Members representing Class "A" Members at a special meeting held for such purpose at least 30 days, but not more than 90 days, prior to expiration of the 15-year period, sote not to extend the 15-year period. Furthermore, if the additional five-year extension occurs under the preceding sentence, the 20-year period for conversion shall be extended an additional three years unless a majority of the Voting Members representing Class "A" Members at a special meeting held for such purpose, at least 30, but not more than 30, days prior to expiration of the 20-year period, vote not to extend the 20 year-(iii) when, in its discretion, the Declarant so determines.

from and after the happening of these events, whichever occurs first, the Class "8" Member shall be deemed to be a Class "A" Member entitled to one wate for each Unit in which it holds the interest required for acaberohip under Section 3.2. The Declarant shall have a right to disapprove actions of the Board and committees as provided in Section 3.18 of the By-Laws.

In the event Class "B" membership has not terminated when more than 50% of the Maximum Units are owned by Retail Owners, the Association shall form a "Transitional Advisory Committee" in accordance with C.R.S. 94.604 and Section 5.3 of the By-Laws. No later than when 75% of the Maximum Units are owned by Retail Owners, the Declarant shall call a meeting for the purpose of turning over administration responsibilities to the Association in accordance with 0.2.5. 94.609 and 94.616.

(c) Brecise of Voting Rights. Except as otherwise specified in this Declaration or the By-Laws, the vote for each Unit owned by a Class "A" Member shall be exercised by Voting Member, as described within Section 1.37 and this Section.

In any situation in which a Member is entitled personally to exercise the vote for his or her Unit and there is more than one Owner of a particular Unit, the rote for such Unit shall be exercised as such co-Owners determine among themselves and advise the secretary of the Association in writing prior to any meeting. Absent such advice, the Unit's vote shall be suspended if more

Article IV RICERS AND OBLIGHTIONS OF THE ASSOCIATION

- 4.1 Compos Area. The Association, subject to the rights of the Owners set forth in this Declaration, shall manage and control the Common Area and all improvements thereon (including, without limitation, furnishings, equipment, and common landscaped areas), and shall keep it in good, clean, attractive, and samitary condition, order, and repair, consistent with this Declaration and
- 4.2 Personal Property and Real Property for Common Use. The Association may acquire, hold, and dispose of tangible and intangible personal property and real property. Declarant may convey to the Association improved or unimproved real estate located within the properties described in Exhibits "A" or "B", personal property and leasehold and other property interests. Such property shall be accepted by the Association and thereafter shall be maintained as Common Area by the Association at its expense for the benefit of its Members, subject to any restrictions set forth in the deed. The Declarant shall convey the initial Common Area to the Association prior to or concurrent with the conveyance of a Unit to a Retail Owner.
- 4.3 Baforcement. The Association may impose sanctions for violations of this Declaration, the By-Laws, or rules in accordance with procedures set forth in the By-Laws, including reasonable monetary fines and suspension of the right to vote and to use any recreational facilities within the Common Area. In addition, in accordance with Section 3.23 of the By-Laws, the Association may exercise self-belp to cure violatious, and may suspend any services it provides to the Unit of any Owner who is more than 30 days delinquent in paying any assessment or other charge due the Association. The Board may seek relief in any court for violation or

to abate nuisances.

The Association, by contract or other agreement, may enforce county ordinances, if applicable, and permit Deschutes County to enforce applicable ordinances on the Properties for the benefit of the Association and its Nembers.

- 4.4 <u>Implied Rights; Board Authority.</u> The Association may exercise any other right or privilege given to it expressly by this Declaration or the By-Laws, or reasonably implied from, or reasonably necessary to effectuate any such right or privilege. Except as otherwise specifically provided in this Declaration, the By-Laws, Articles, or by law, all rights and powers of the Association may be exercised by the Board without a vote of the newbership.
- 4.5 Governmental Interests. So long as the Declarant owns any property described on Exhibits "A" or "B", the Declarant may designate sites within the Properties for fire, police, utility facilities, public schools and parks, and other public facilities. The sites way include Common Areas.
- 4.6 <u>Indemnification</u>. The Association shall indemnify every officer, director, and committee member against all expenses, including counsel fees, reasonably incurred in connection with any action, suit, or other proceeding (including settlement of any director, or committee member.

The officers, directors, and committee members shall not be liable for any mistake of judgment, negligent or otherwise, except for their own individual willful misfeasance, unifeasance, nisconduct, or bad faith. The officers and directors shall have no personal liability with respect to any contract or other commitment made or action taken in good faith on behalf of the Association (except to the extent that such officers or directors may also be Members of the Association). The association shall indemnify and forever hold each such officer, director and committee member harmless from any and all liability to others on account of any such contract, or former officer, director, or committee members may be entitled. The Association shall, as a Common Expense, maintain adequate general liability and officers' and directors' liability insurance to fund this obligation, if such insurance is reasonably available.

Pursuant to O.R.S. 65.784, if the Association indemnifies or advances expenses pursuant to this Section or O.R.S. 65.391-65.401, of the Association shall report the indemnification or advance in writing to the Members with or before the notice of the next meeting of the Members.

- 4.7 <u>Dedication of Common Areas</u>. The Association may dedicate portions of the Common Areas to Deschutes County, Oregon or to any other local, state, or federal governmental entity, subject to such approval as may be required by Section 14.2.
- 4.8 Secrity. The Association may, but shall not be obligated to, maintain of support certain activities within the Properties designed to make the Properties safer than they otherwise might be. Belter the Association, the declarant, hor any successor for any seall in any ray be considered insurers or guarantors by security rithin the properties, hor shall any of them be beind liable representation or marranty is made that any pire protection system, burglar alarm system or other security system caphol be compromised or circumptrated, nor that any successors or security measures undertaken whill in a case prevent cose or provide the detection or protection for which the system is designed or intended. Each owner acknowledges, understands and coverants to inform its terants that the association, its board of directors and committees, declarant, and any successor declarant are not insurers and teat back access or their parties.

Article V MAINTENANCE

5.1 <u>Association's Responsibility.</u> The Association shall maintain and keep in good repair the Area of Common Responsibility, which shall include, but need not be limited to:

- (a) all landscaping and other flora, parks, signage, structures, and improvements, including any private streets, bike and pedestrian pathways/trails, situated upon the Common Area;
- (b) landscaping, sidewalks, street lights and signage within public rights-of-way within or abutting the Properties, and landscaping and other flora within any public utility easements and conservation easements within the Properties (subject to the terms of any easement agreement relating thereto); provided, however, that it shall be the responsibility of each Owner to landscape, irrigate, and maintain any area within public rights of way between such Owner's Unit and the paved coadway located in rights of way adjacent to such Owner's Unit from the date a certificate of occupancy is issued with respect to improvements on such Unit.
- (c) such portions of any additional property included within any Area of Common Responsibility as may be distated by this Declaration, any Supplemental Declaration, or any contract or agreement for maintenance thereof entered into by the Association; and
- (d) any property and facilities owned by the Declarant and made available, on a temporary or permanent basis, for the primary use and enjoyment of the Association and its Members, such property and facilities to be identified by written notice from the Declarant to the Association and to remain a part of the Area of Gommon Responsibility and he maintained by the Association until such time as Declarant revokes such privilege of use and enjoyment by written notice to the Association.

There are hereby reserved to the Association, easements over the Properties as necessary to enable the Association to fulfill such responsibilities. The Association shall maintain the facilities and equipment within the Area of Common Responsibility in continuous operation, except for reasonable periods as necessary to perform required maintenance or repairs, unless Voting Members representing 75% of the Class "A" votes and the Declarant, as long as the Declarant owns any property described on Exhibits "A" or "B", agree in writing to discontinue such operation.

The Association may maintain other property which it does not own, including without limitation publicly owned property, conservation easements held by monprofit entities, and other property dedicated to public use, if the Board determines that sach maintenance is necessary or desirable to maintain the Community-Wide Standard.

Bicept as otherwise specifically provided berein, all costs associated with maintenance, repair and replacement of the Area of Common Responsibility shall be a common expense to be allocated among all Units in the manner of and as part of the Base Assessment, without prejudice to the right of the Association to seek reimbursement from the Gwner(s) of, or other Person responsible for, certain portions of the Area of Common Responsibility pursuant to this Declaration, other recorded covenants, or agreements with the Owner(s) thereof.

- 5.2 Owner's Responsibility. Each Owner shall maintain his or her Unit and all structures, parking areas, landscaping, and other improvements comprising the Unit in a manner consistent with the Community-Wide Standard and all applicable covenants unless such maintenance responsibility is otherwise assumed by or assigned to the Association pursuant to any Supplemental Declaration or other declaration of covenants applicable to such Unit. In addition to any other enforcement rights, if an Owner fails properly to perform his or her maintenance responsibility, the Association may perform such maintenance responsibilities and assess all costs incurred by the Association against the unit and the Owner in accordance with Section 10.7(b). The Association shall afford the Owner reasonable notice and an opportunity to cure the problem prior to entry, except when entry is required due to an emergency situation.
- 5.3 Standard of Performance. Maintenance, as used in this Article, shall include, without limitation, repair and replacement as needed, as well as such other daties, which may include irrigation, as the Board may determine necessary or appropriate to satisfy the Community-Wide Standard. All maintenance shall be performed in a manner consistent with the Community-Wide Standard and all applicable covenants.

Motwithstanding anything to the contrary contained herein, the Association, and/or an Owner, shall not be liable for property damage or personal injury occurring on, or arising out of the condition of, property which it does not own unless and only to the extent that it has been negligent in the performance of its maintenance responsibilities.

5.4 Party Walls and Similar Structures.

- (a) General Rules of Law to Apply. Back wall, fence, driveway or similar structure built as a part of the original construction on the Units which serves and/or separates any two adjoining Units shall constitute a party structure. To the extent not inconsistent with the provisions of this Section, the general rules of law regarding party walls and liability for property
- (b) Sharing of Repair and Maintenance. All Owners who make use of the party structure shall share the cost of reasonable repair and maintenance of such structure equally.
- (c) Danage and Destruction. If a party structure is destroyed or danaged by fire or other casualty, then to the extent that such damage is not corered by insurance and repaired out of the proceeds of insurance, any Owner who has used the structure may restore it. If other Owners subsequently use the structure, they shall contribute to the restoration cost in equal proportions. However, such contribution will not prejudice the right to call for a larger contribution from the other users under any rule of law
- (d) Right to Contribution Runs With Land. The right of an Owner to contribution from any other Owner under this Section shall be appurtenant to the land and shall pass to such Owner's successors-in-title.

Article 91 INSURANCE AND CASUALTY LOSSES

6.1 Association Insurance. The Association, acting through its Board or its duly authorized agent, shall obtain blanket "allrist property issurance, if reasonably available, for all insurable improvements on the Common Area and on other portions of the Area of Common Responsibility to the extent that it has assumed responsibility for maintenance, repair and/or replacement to the event of a loss. The Association shall have the authority to and interest in insuring any privately or publicly owned property for which the Association has maintenance or repair responsibility. Such property shall include, by way of illustration and not limitation, any insurable improvements on or related to parks, rights-of-ways, medians, easements, and walkways which the Association is obligated to maintain. If blanket "all-risk" coverage is not generally available at reasonable cost, then the Association shall obtain fire and extended coverage, including coverage for vandalism and malicious mischief. The face amount of the policy shall be sufficient to cover the full replacement cost of the insured property. The cost of such insurance shall be a Common Expense to be allocated among all Units subject to assessment as part of the annual Base Assessment.

The Association shall have no insurance responsibility for any part of any Private Amenity property.

The Association also shall obtain a public liability policy on the Area of Common Responsibility, insuring the Association and its Members for damage or injury caused by the negligence of the Association or any of its Members, employees, agents, or contractors while acting on its behalf. If generally available at reasonable cost, the public liability policy shall have at least a \$1,000,0000.00 combined single limit as respects bodily injury and property damage and at least a \$3,000,000.00 limit per occurrence

Premiums for all insurance on the Area of Common Responsibility shall be Common Expenses and shall be included in the Ease Assessment.

The policies may contain a reasonable deductible which shall not be subtracted from the face amount of the policy in determining whether the insurance at least equals the required coverage. In the event of an insured loss, the deductible shall be treated as a Common Expense in the same manner as the premiums for the applicable insurance coverage. However, if the Board reasonably determines, after notice and an opportunity to be heard to accordance with Section 3.23 of the By-Laws, that the loss is the result of the negligence or willful conduct of one or more Owners or occupants, then the Board may specifically assess the full amount of such deductible against the Unit of such Owner or occupant, pursuant to Section $10.7(\delta)$. All insurance coverage obtained by the Association shall:

- (a) be written with a company authorized to do business in Oregon which holds a Best's rating of A or better and is assigned a financial size category of IX or larger as established by A.M. Best Company, Inc., if reasonably available, or, if not available, the most nearly equivalent rating which is available;
- (b) be written in the name of the Association as trustee for the benefitted parties. Policies on the Common Area shall be for the benefit of the Association and its Members;
- (c) rest in the Board, exclusive authority to adjust losses; provided, however, no Mortgagee having an interest in such losses may be prohibited from participating in the settlement negotiations, if any, related to the loss;
- (d) not be brought into contribution with insurance purchased by individual Owners, orcupauls, or their Hortgagees;
 and
- (e) have an inflation guard endorsement, if reasonably available. If the policy contains a co-insurance clause, it shall also have an agreed amount endorsement. The Association shall arrange for an annual review of the sufficiency of insurance coverage by our or more qualified persons, at least one of whom must be in the real estate industry and familiar with construction in the Deschutes County, Oregon area.

The Board shall use reasonable efforts to secure insurance policies containing endorsements that:

- (a) waive subrogation as to any claims against the Association's Board, officers, employees, and its manager, the Owners and their tenants, servants, agents, and guests;
 - (b) waive the insurer's rights to repair and reconstruct instead of paying cash;
- (c) preclude cancellation, invalidation, suspension, or nonrenewal by the insurer on account of any one of more individual Owners, or on account of any curable defect of violation without prior written demand to the Association to cure the defect or violation and allowance of a reasonable time to cure;
 - (d) exclude individual Owners' policies from consideration under any "other insurance" clause; and
- (e) require at least 30 days prior written notice to the Association of any cancellation, substantial modification, or nonrenewal.

The Association shall also obtain, as a Common Expense, worker's compensation insurance and employer's liability insurance, if and to the extent required by law; directors' and officers' liability coverage, if reasonably available, and flood insurance, if advisable.

The Association also shall obtain, as a Common Expense, a fidelity bond or bonds, if generally available at reasonable cost, covering all persons responsible for handling association funds. The Board shall determine the amount of fidelity coverage in its best business judgment but, if reasonably available, shall secure coverage equal to not less than one-sixth of the amount Base Assessments on all Units plus reserves on band. Bonds shall contain a maiver of all defenses based upon the exclusion of persons serving without compensation and shall require at least 30 days prior notice to the Association of any cancellation, substantial modification, or nonrenewal.

6.2 Owners Insurance. By virtue of taking title to a Unit, each Owner covenants and agrees with all other Owners and with the Association to carry blanket "all-risk" property insurance on its Unit(s) and structures thereon providing full replacement cost coverage less a reasonable deductible, unless the Association carries such insurance (which they are not obligated to do bereunder).

Bach Owner further covenants and agrees that in the event of damage to, or destruction of, structures on or comprising his or her Unit, he or she shall proceed promptly to repair or to reconstruct in a manner consistent with the original construction or such other plans and specifications as are approved in accordance with Article XI of this Declaration. Alternatively, the Owner shall

clear the Unit of all debris and ruins and maintain the Unit in a neat and attractive, landscaped condition consistent with the Community-Nide Standard. The Owner shall pay any costs which are not covered by insurance proceeds.

6.3 Damage and Destruction.

- (a) Immediately after damage or destruction to all or any part of the Properties covered by insurance written in the name of the Association, the Board or its duly authorized agent shall file and adjust all insurance claims and obtain reliable and detailed estimates of the cost of repair or reconstruction. Repair or reconstruction, as used in this paragraph, means repairing or restoring the property to substantially the condition to which it existed prior to the damage, allowing for changes or improvements necessitated by changes in applicable building codes.
- (b) Any damage to or destruction of the Common Arra shall be repaired or reconstructed unless the Members representing at least 75% of the total class "A" rotes and the Declarant, as long as the Declarant owns any property described in Exhibits "A" or "B" of the Declaration, decide within 60 days after the loss not to repair or reconstruct. No Mortgagee shall have the right to participate in the determination of whether the damage or destruction to the Common Area shall be repaired or reconstructed.

If either the incurance proceeds or reliable and detailed estimates of the cost of repair or reconstruction, or both, are not available to the Association within such 60-day period, then the period shall be extended until such funds or information are available. Bowever, such extension shall not exceed 60 additional days.

- (r) If determined in the manner described above that the damage or destruction to the Common Area shall not be repaired or reconstructed and no alternative improvements are authorized, the affected property shall be cleared of all debris and maintained by the Association in a neat and attractive, landscaped condition consistent with the Community-Nide Standard.
- 6.4 <u>Disbursements of Proceeds</u>. Any insurance proceeds remaining after paying the costs of repair or reconstruction, or after such settlement as is necessary and appropriate, shall be retained by and for the benefit of the Association and placed in a capital improvements account. This is a covenant for the benefit of Mortgagees and may be enforced by the Mortgagee of any affected Unit.
- 6.5 Repair and Reconstruction. If insurance proceeds are insufficient to cover the costs of repair or reconstruction, the Board shall, without a vote of the Voting Members, levy Special Assessments against those Unit Owners responsible for the premiums for the applicable insurance coverage under Section 6.1.

Article VII NO PARTITION

Except as permitted in this Declaration, there shall be no judicial partition of the Common Area. No Person shall seek any judicial partition unless the Properties or such portion thereof have been removed from the provisions of this Declaration. This Article shall not prohibit the Board from acquiring and disposing of tangible personal property nor from acquiring and disposing of real property which may or may not be subject to this Declaration.

Article VIII CONDEMNATION

If any part of the Common Area shall be taken (or conveyed in lieu of and under threat of condemnation by the Board acting on the written directice of Toting Members representing at least 66 2/3% of the total Class "A" votes in the Association, and of the Beclarant, as long as the Declarant owns any property described on Exhibits "A" or "B" of the Declaration) by any authority having the power of condemnation or eminent domain, each Owner shall be entitled to written notice. The award made for such taking shall be payable to the Association as trustee for all Owners to be disbursed as follows:

If the taking involves a portion of the Common Area on which improvements have been constructed, the Association shall restore or replace such improvements on the remaining land included in the Common Area to the extent available, unless within 60 days after

such taking, the Declarant, so long as the Declarant owns any property described in Exhibits "A" or "B" of this Declaration, and Yoting Bembers representing at least 75% of the total Class "A" votes in the Association shall otherwise agree. Any such construction shall be in accordance with plans approved by the Board. The provisions of Section 6.4 and 6.5 regarding funds for the repair of disage or destruction shall apply.

If the taking does not involve any improvements on the Common Area, or if a decision is made not to repair or restore, or if net funds remain after any such restoration or replacement is complete, then such award or net funds shall be disbursed to the Association and used for such purposes as the Board shall determine.

Article IX AMMEXATION AND WITHDRAWAL OF PROPERTY

9.1 <u>Appearation Without Approval of Membership.</u> Prior to the sale of the first Unit to a Retail Owner, Declarant may unilaterally subject to the provisions of this Declaration all or portions of the real property described in Exhibit "B" until all property described on Exhibit "B" until all property described on Exhibit "B" bas been subjected to this Declaration or 30 years after the recording of this Declaration, whichever is earlier. After the sale of the first Obit to a Retail Owner, Declarant may unilaterally subject to the provisions of this Declaration all or any portion of the real property described in exhibit "B", until the earlier of (a) the date on which all property described on Exhibit "B" has been subjected to this Declaration, or (b) the later of (i) four years from the most recent sale to a Retail Owner, or (ii) four years from the date of recording of the most recent Supplemental Declaration to aggree additional property.

Declarant may transfer or assign this right to annex property, provided that the transferee or assignee is the developer of at least a portion of the real property described in Babibits "A" or "B" and that such transfer is memorialized in a written, recorded instrument executed by Declarant. Wothing in this Declaration shall be construed to require the Declarant or any successor to annex or develop any of the property set forth in Exhibit "B" in any manner whatsoever.

Such annexation shall be accomplished by filing a Supplemental Declaration in the Office of the County Clerk describing the property to be assessed and specifically subjecting it to the terms of this Declaration. Such Supplemental Declaration shall not require the consent of Members, but shall require the consent of the Owner of such property, if other than Declarant. Any such assessable be effective upon the filing for record of such Supplemental Declaration unless otherwise provided therein.

9.2 <u>Annexation With Approval of Membership.</u> The Association or the Declarant may subject any real property to the provisions of this Declaration with the consent of the owner of such property, the affirmative vote of Yoting Members representing 66 2/3% of the Class "A" votes of the Association represented at the meeting duly called for such purpose, and the consent of the Declarant so long as Declarant owns property subject to this Declaration or which may become subject to this Declaration in accordance with Section 9.1.

Such annexation shall be accomplished by filing a Supplemental Declaration in the Office of the County Clerk describing the property to be annexed and specifically subjecting it to the terms of this Declaration. Any such Supplemental Declaration shall be signed by the President and the Secretary of the Association, and by the owner of the annexed property. Any such appearation shall be effective upon filing unless otherwise provided therein.

- 9.3 <u>Withdrawal of Property.</u> The Declarant reserves the right to amend this Declaration so long as it has a right to amen additional property pursuant to this Article, without prior notice and without the consent of any Person, for the purpose of removing property then owned by the Declarant, its affiliates, or the Association from the coverage of this Declaration, to the extent originally included in error or as a result of any changes in the Declarant's plans for the Properties, provided such withdrawal is not unequivocally contrary to the overall, uniform scheme of development for the Properties.
- 9.4 Additional Covenants and Easements. The Declarant may unilaterally subject any portion of the property submitted to this Declaration initially or by Supplemental Declaration to additional covenants and easements, including covenants obligating the Association to maintain and insure such property on behalf of the Owners and obligating such Owners to pay the costs incurred by the Association. Such additional covenants and easements shall be set forth in a Supplemental Declaration filed wither concurrent with or after the appearation of the subject property, and shall require the written consent of the owner(s) of such property, if other

than the Declarant.

9.5 Amendment. This Article shall not be amended without the prior written consent of Declarant so long as the Declarant owns any property describe in Exhibits "A" or "B".

Article X ASSESSMENTS

10.1 <u>Creation of Assessments.</u> The Association is hereby authorized to levy assessments against each Unit for Association expenses as the Board may specifically authorize from time to time. There shall be three types of assessments for Association expenses:

(a) Base Assessments to fund Common Expenses for the general benefit of all units; (b) Special Assessments as described in Section 10.6; and (c) Specific Assessments as described in Section 10.7. Each Owner, by accepting a deed or entering into recorded contract of sale for any portion of the Properties is deemed to covenant and agree to pay these assessments.

All assessments (except as otherwise provided in Section 10.7(b)), together with interest from the due date of such assessment at a rate determined by the Association (not to exceed the highest rate allowed by Oregon Law), late charges, costs, and reasonable attorney fees, shall be a charge and continuing lien upon each Unit which the assessment is made until paid, as more particularly provided in Section 10.10. Each such assessment, together with interest, late charges, costs, and reasonable attorney fees, also shall be the personal obligation of the Person who was the Owner of such Unit at the time the assessment arose. Upon a transfer of title to a Unit, the grantee shall not be liable for any assessments and other charges due at the time of conveyance unless expressly assumed by him or her. No first Mortgages who obtains title to a Unit by exercising the remedies provided in its Mortgage shall be liable for unpaid assessments which accrued prior to such acquisition of title.

Assessments shall be paid in such manner and on such dates as the Board may establish. If the Board so elects, assessments may be paid in two or more installments. Unless the Board otherwise provides, the Base Assessment shall be due and payable in advance on the first day of each fiscal year. If any Owner is delinquent in paying any assessments or other charges levied on his or her Unit, the Board may require any unpaid installments of all outstanding assessments to be paid in full immediately.

The Association shall, upon request, furnish to any Owner liable for any type of assessment a certificate in writing signed by an officer of the Association setting forth whether such assessment has been paid. Such certificate shall be conclusive evidence of payment. The Association may require the advance payment of a reasonable processing fee for the issuance of such certificate.

No Owner may exempt himself or herself from liability for assessments, by monuse of Common Area, abandonment of his or her dain, or any other means. The obligation to pay assessments is a separate and independent covenant on the part of each Owner. Mo diminution or abatement of assessments or set-off shall be claimed or allowed for any alleged failure of the Association or Board to take some action or perform some function required of it, or for incommended or disconfort arising from the making of repairs or improvements, or from any other action it takes.

After the commencement of assessment payments as to any Unit, Declarant, if any, covenants and agrees to pay the full amount of any applicable assessment for each occupied Unit it owns; notwithstanding anything contained berein to the contrary, the Declarant shall not be required to pay Base, Special, or Specific Assessments for unoccupied units that it owns.

Motwithstanding anything to the contrary herein, the Declarant may contribute assessments due from it in services and/or materials ("In-Rind"), rather than in money. The amount by which monetary assessments shall be decreased as a result of any In-Rind contribution shall be the fair market value of the contribution. If the Declarant and the Association disagree as to the value of any contribution, the Declarant shail supply the Association with a detailed explanation of the service performed and material furnished, and the Association shall acquire bids for performing like services and furnishing like materials from three (3) independent contractors approved by the Declarant who are in the business of providing such services and materials. If the Association and the Declarant are still unable to agree on the value of the contribution, the value shall be deemed to be the average of the bids received from the independent contractors.

10.2 Capitalization of Association. Upon acquisition of record title to each Unit by its first Retail Owner, a contribution

shall be made by, or on behalf of, the purchaser to the working capital of the Association in an amount equal to one-sizth of the annual Base Assessment per Unit for that year. For purposes of this Section, the term "Base Assessment shall be deemed to include the amount assessed against each Unit plus a pro-rate allocation of any amounts (values) the Association received through any subsidy, maintenance agreement, or In-Kind services or materials, if any. This amount shall be in addition to, not in lieu of, the annual Base Assessment and shall not be considered an advance payment of such assessment. This amount shall be deposited into the purchase and sales estrow and disbursed therefrom to the Association for use in covering operation expenses and other expenses incurred by the Association pursuant to the terms of this Declaration and the By-Laws.

- 10.3 Date of Conmencement of Assessments. Subject to Section 10.1, the obligation to pay the assessments provided for herein shall commence as to all Units on the earlier of: (a) the first day of the month following the first corresponds of a Unit to a Retail Owner; or (b) the first day of the month following conveyance of the Common Area to the Association. The first annual Assessment shall be adjusted according to the number of days remaining in the fiscal year at the time assessments commence on the Unit.
- 10.4 <u>Comportation of Base Assessment.</u> The Board shall prepare a budget covering the estimated Common Expenses during the coming year pursuant to the terms and provisions set forth in Section 3.20(f)(i) of the By-Laws. The budget shall include a capital contribution to establish a reserve fund in accordance with a budget separately prepared as provided in Section 10.5

The Base Assessment shall be levied equally against all Units subject to assessment and shall be set at a level which is reasonably expected to produce total income for the Association equal to the total budgeted Common Bypenses, including reserves. In determining the level of assessments, the Board, in its discretion, may consider other sources of funds available to the Association. In addition, the Board shall take into account the number of Units subject to assessment under Section 10.7(b) on the first day of the fiscal year for which the budget is prepared and the number of Units reasonably anticipated to become subject to assessment during the fiscal year.

The Board shall send a copy of the budget and notice of the amount of the Base Assessment for the following year to each Owner not less than 45 nor more than 60 days prior to the beginning of the fiscal year for which it is to be effective.

- 10.5 Reserve Budget and Capital Contribution. The Board shall annually prepare a reserve budget which takes into account the number and nature of replaceable assets, the expected life of each asset, and the expected repair or replacement cost and in accordance with O.R.S. 94.595. The Board shall set the required capital contribution in an amount sufficient to permit meeting the projected needs of the Association, as shown on the budget, with respect both to amount and timing by annual Base Assessments over the budget period.
- 10.6 Special Assessments. In addition to other authorized assessments, the Association may levy Special Assessments from time to time to cover unbudgeted expenses or expenses in excess of those budgeted subject to the limitations set forth in Section 10.8. Such Special Assessment may be levied against the entire membership, if such Special Assessment is for common expenses. Special Assessments shall be payable in such manner and at such times as determined by the Board, and may be payable in installments extending beyond the fiscal year in which the Special Assessment is approved.
- 10.7 <u>Specific Assessments.</u> The Board shall have the power to levy Specific Assessments against a particular Unit or Units constituting less than all Units within the Properties as follows:
- (a) to cover the costs, including overhead and administrative costs, of providing benefits, items, or services to the Unit or occupants thereof upon request of the Owner pursuant to a menu of special services which the Board may from time to time authorize to be offered to Owners (which might include, without limitation, landscape maintenance, handyman service, pool cleaning, pest control, etc.), which assessments may be levied in advance of the provision of the requested benefit, item or service as a deposit against charges to be incurred by the Owner; and
- (b) to cover costs incurred in bringing the Unit into compliance with the terms of this Declaration, any applicable Supplemental Declaration, the By-Laws or rules, or costs incurred as a consequence of the conduct of the Owner or occupants of the Unit, their licensees, invitees, or guests; provided, the Board shall give the Unit Owner prior written notice and an opportunity for a hearing before levying a Specific Assessment under this subsection (b).

10.8 Limitation of lucreases of Assessments. Notwithstanding any provision to the contrary, and except for assessment increases necessary for emergency situations or to reimburse the Association pursuant to Section 10.7, the Board may not impose a Base Assessment or Specific Assessment that is more than 20% greater than each of those assessments for the immediately preceding fiscal year nor a majority vote of a quorum of Voting Members representing the Nembers which are subject to the applicable assessment at a meeting of the Association. For purposes of this Section, "quorum" means the Foting Members representing more than 50% of the Members which are subject to the applicable assessment at a meeting are subject to the applicable assessment. For purposes of this Section, the term "Base Assessment" shall be deemed to include the amount assessed against each Unit plus a pro-rata allocation of any amounts the Association received through any maintenance agreement, if any, in effect for the year immediately preceding the year for which the assessment is to be increased.

An emergency situation is any one or more of the following:

- (a) an extraordinary expense required by an order of a court;
- (b) an extraordinary expense necessary to repair or maintain the Properties or any part of them for which the Association is responsible where a threat to personal safety on the Properties is discovered; or
- (c) an extraordinary expense necessar; to repair or maintain the Properties or any part of them for which the Association is responsible which could not have been reasonably foreseen by the Board in preparing and distributing the proformate pursuant to Section 10.4. Bowever, prior to the imposition or collection of such an assessment, the Board shall pass a resolution containing written findings as to the necessity of the extraordinary expenses involved and why the expense was not or could not have been reasonably foreseen in the budgeting process. Such resolution shall be distributed to the Members with the notice
- 10.9 <u>Failure to Assess.</u> Pailure of the Board to fix assessment amounts or rates or to deliver or usil to each owner an assessment notice shall not be deemed a waiver, modification, or a release of any Owner from the obligation to pay assessment. In such event, each Owner shall continue to pay Base Assessments on the same basis as for the last year for which an assessment was made, if any, until a new assessment is made, at which time the Association may retroactively assess any shortfalls in collections.
- 10.10 Lien for Assessments. All assessments authorized in this Article shall constitute a lieo against the Unit against which they are levied until paid, unless otherwise specifically precluded in this Declaration. The lien shall also secure payment of interest, late charges (subject to the limitation of Oregon law), and costs of collection (including attorney fees). Such lien shall be superior to all other liens, except (a) the liens of all taxes, bonds, assessments, and other levies which by law would be superior, and (b) the lien or charge of any first Mortgage of record (meaning any recorded Mortgage with first priority over other mortgages) wade in good faith and for value. The Association may enforce such lien, when delinquent, by suit, judgment, and foreclosure. The lien shall be effective for a period of ten (10) years from the date of filing. The lien shall be filed with the "Office of the County Clert". The lien shall be foreclosed pursuant to Chapter 88 or O.R.S.

The Association may bid for the Unit at the foreclosure sale and acquire, hold, lease, mortgage, and convey the Unit. While a Unit is owned by the Association following foreclosure: (a) no right to vote shall be exercised on its behalf; (b) no assessment shall be levied on it; and (c) each other Unit shall be charged, in addition to its usual assessment, its equal pro rata share of the assessment that would have been charged such Unit had it not been acquired by the Association. The Association may sue for unpaid Common Expenses and costs without foreclosing or waiving the lien securing the same.

The sale or transfer of any Unit shall not affect the assessment lien or relieve such Unit from the lien for any subsequent assessment. However, the sale or transfer of any Unit pursuant to foreclosure of the first Mortgage shall extinguish the lien as to any installments of such assessments due prior to such sale or transfer. A Mortgage or other purchaser of a Unit who obtains of title pursuant to foreclosure of the Mortgage shall not be personally liable for assessments on such Unit due prior to such acquisition of title.

10.11 Exempt Property. The following property shall be exempt from payment of Base Assessments and Special Assessments:

- (a) All common Areas:
- (b) Any property dedicated to and accepted by any governmental authority or public utility; and
- (c) Unoccupied lots owned by Declarant.

In addition, the Declarant shall have the right, but not the obligation, to grant exemptions to certain Persons qualifying for Section 501(c) status under the Internal Revenue Gode so long as such Persons own property subject to this Declaration for purposes listed in Section 501(c).

Article XI ARCHITECTURAL STANDARDS

11.1 General. No structure shall be placed, erected, or installed upon any Unit, and no improvements (including staking, clearing, excavation, grading and other site work, exterior alteration of existing improvements, planting or removal of landscaping materials, and installation or removal of an irrigation system) shall take place except in compliance with this Article and the Design Guidelines and upon approval of the appropriate committee under Section 11.2.

Any Owner may remodel, paint or redecorate the interior of structures on his or her Unit without approval. However, modifications to the interior of screened porches, patios, and similar portions of a Unit visible from outside the structures on the Unit shall be subject to approval. No approval shall be required to repaint the exterior of a structure in accordance with the originally approved color scheme or to rebuild in accordance with originally approved plans and specifications.

All dwellings and landscaping constructed or placed on any portion of the Properties shall be designed by and built in accordance with the plans and specifications of a licensed architect or licensed building designer.

This article shall not apply to the activities of the Declarant, nor to improvements to the Common Area by, or on behalf of the Association.

This Article may not be amended without the Declarant's written consent so long as the Declarant owns any land subject to this Declaration or subject to accuration to this Declaration.

11.2 Architectural Review. Responsibility for administration of the Design Guidelines, as defined below, and review of all applications for construction and modifications under this Article shall be handled by the committee as described herein. The Members of the committee need not be members of the Association or representatives of Members, and may, but need not, include architects, engineers or similar professionals, whose compensation, if any, shall be established from time to time by the Board. The Board may establish and charge reasonable fees for review of applications hereunder and may require such fees to be paid in full prior to review.

The Design Review Committee (DRC) shall consist of at least three, but not more than five persons. So long as Declarant, any affiliate of the Declarant, or any Builder owns any Unit primarily for development and/or sale, the Declarant retains the right to appoint all members of the BRC, who shall serve at the discretion of the Declarant. There shall be no surrender of this right prior to that time, except in a written instrument in recordable form executed by Declarant. Opon surrender of such right, the Board of Directors may appoint the members of the DRC, who shall serve and may be removed at the discretion of the Board of Directors.

The DRC shall have exclusive jurisdiction over all original construction on or any modifications, additions, or alterations made to any Units owned by Declarant, any affiliates of Declarant, any Boilder, or Units owned by Retail Owners.

11.3 <u>Guidelines and Procedures.</u> The Declarant shall prepare the initial design and development guidelines and application and review procedures ("Design Guidelines") which shall apply to all construction activities within the Properties. The Design Guidelines may contain general provisions applicable to all of the Properties, as well as specific provisions which vary from one portion of the Properties to another depending upon the location, unique characteristics, and intended use.

Barrier Likewick

The DRC shall adopt such Design Guidelines at its initial organizational meeting and thereafter shall have sole and full authority to amend them. Any amendments to the Design Guidelines shall apply to construction and modifications commenced after the date of such amendment only and shall not apply to require modifications to or removal of structures previously approved once the approved construction or modification has commenced.

The DRC shall make the Design Guidelines available to Owners and Builders who seek to engage in development or construction within the Properties and all such Persons shall conduct their activities in accordance with such Design Guidelines. At the Declarant's discretion, such Design Guidelines may be recorded in the Office of the County Clerk, in which event the recorded version, as it may unliaterally be amended from time to time, shall control in the event of any dispute as to which version of the Design Guidelines was in effect at any particular time.

11.4 Submission of Plans and Specifications.

- (a) No construction or improvements shall be commenced, erected, placed or maintained on any Unit, nor shall any design, exterior addition, change or alteration be made thereto, until the plans and specifications ("Plans") showing site layout, structural design, exterior elevations, exterior materials and colors, signs, landscaping, drainage, lighting, irrigation, utility facilities layout, and screening therefor shall have been submitted to and approved in writing by the DRC. The Design Guidelines shall set forth the procedure for submission of the Plans.
- (b) In reviewing each submission, the DRC may consider visual and environmental impact, ecological compatibility, natural platforms and finish grade elevation, harmony of external design with surrounding structures and environment, and location in relation to surrounding structures and plant life. The committee may require relocation of native plants within the construction site or the installation of an irrigation system for the landscaping, including the natural plant life on the Unit, as condition of approval of any submission.

The BPC shall, within 45 days after receipt of each submission of the Plans, advise the party submitting the same in writing at an address specified by such party at the time of submission of (i) the approval of Plans, or (ii) the segments or features of the Plans which are deemed by such committee to be inconsistent or not in conformity with the Declaration and/or the Design Guidelines, the reason for such finding and suggestions for the coring of such objections. In the erent the committee fails to advise the submitting party by written notice within the time set forth above, of either the approval or disapproval of the Plans, approval shall be deemed to have been given. Notice shall be deemed to bave been given at the time the envelope containing such notice, properly addressed, and postage prepaid, is deposited with the U.S. Postal Service, registered or certified mail, return receipt requested. Personal delivery of such written notice shall, however, be sufficient and shall be deemed to have been given at the time of delivery.

- (c) If construction does not commence on a project for which Plans have been approved within 12 months of such approval shall be deemed withdrawn and it shall be necessary to resubmit the Plans to the DRG for reconsideration.
- 11.5 No Naiver of Puture Approvals. Each Owner acknowledges that the members of the DRC will change from time to time and that interpretation, application and enforcement of the Design Guidelines may vary accordingly. Approval of proposals, plans and specifications, or drawings for any work done or proposed, or in connection with any other matter requiring approval, shall not be deemed to constitute a maiver of the right to withhold approval as to any similar proposals, plans and specifications, drawings, or other matters subsequently or additionally submitted for approval.
- 11.5 <u>Variance.</u> The DRC may authorize variances from compliance with any of its guidelines and procedures when circumstances such as topography, natural obstructions, bardship, or aesthetic or environmental considerations require, but only in accordance with dely adopted rules and regulations. Such variances may only be granted, however, when unique circumstances dictate and no variance shall (a) be effective nuless in writing; (b) be contrary to this Declaration; or (c) estop the DRC from denying a variance in other circumstances. For purposes of this Section, the inability to obtain approval of any governmental agency, the issuance of any permit, the cost of compliance, or the terms of any financing shall not be considered a hardship warranting a variance.
- 11.7 <u>Limitation of Liability</u>. Review and approval of any application pursuant to this Article is made on the basis of anotheric considerations only and the DRC shall not bear any responsibility for easuring the structural integrity or soundness of approved construction or modifications, nor for ensuring compliance with building codes and other governmental requirements. Neither the

Declarant, the Association, the Board, any committee, or member of any of the foregoing shall be held flable for any injury, damages, or loss arising out of the manner or quality of approved construction on, or modifications to, any Unit.

11.8 <u>Baforcement.</u> Any structure or improvement placed or made in violation of this Article shall be deemed to be announforming. Upon written request from the Board or the Declarant, Owners shall, at their own rost and expense, remove such structure or improvement and restore the land to substantially the same condition as existed prior to the nonconforming work. Should an Owner fail to remove and restore as required, the Board or its designees shall have the right to enter the property, remove the violation, and restore the property to substantially the same condition as previously existed. All costs, together with the interest at the maximum rate than allowed by law, may be assessed against the benefited Unit and collected as a Specific Assessment unless otherwise prohibited in this Declaration.

Any contractor, subcontractor, agent, employee, or other invitee of an Owner who fails to comply with the terms and provisions of this Article and the Design Guidelines may be excluded by the Board from the Properties, subject to the notice and hearing procedures contained in the By-Laws. In such event, neither the Association, its officers, or directors shall be held liable to any Person for exercising the rights granted by this paragraph.

In addition to the foregoing, the Association shall have the authority and standing to pursue all legal and equitable remedies available to enforce the provisions of this Activie and the decision of the DRC.

11.9 Building Site. The DRC may establish building areas for each Unit. If so established, construction of a residence shall be restricted to said area, unless the DRC, upon good cause shown by the Unit Owner, shall select another suitable building area upon a Unit. Constructing on such alternate site shall not interfera with the view of any adjacent Unit nor shall it encroach upon any established Unit setback requirement.

Article XII USE GUIDBLINES AND RESTRICTIONS

IZ.1 Plan of Development; Applicability; Effect. Declarant has created ASPEN LAKES as a residential and recreational development and, in furtherance of its and every other Owner's interests, has established a general plan of development for ASPEN LAKES as a master planned community. The Properties are subject to land development, architectural, and Design Guidelines as set forth in upon the properties are subject to guidelines and restrictions governing land use, individual conduct, and uses of or actions upon the properties as provided in this Article XII. This Declaration and resolutions of the Board may establish affirmative and negative covenants, easements, and restrictions ("Use Guidelines and Restrictions").

All provisions of this Declaration and of any Association rules shall also apply to all occupants, tenants, guests, and invitees of any Unit. Any lease on any Unit shall provide that the lessee and all occupants of the leased Unit shall be bound by the terms of this Declaration, the By-Laws, and the rules of the Association.

Declarant promulgates ASPEN LAKES' general plan of development in order to protect all Owners' quality of life and collective interests, the aesthetics and environment within the Properties, and the vitality of and sense of community within ASPEN LAKES all subject to the Board's and the Members' ability to respond to changes in circumstances, conditions, needs, and desires within the master planned community.

Declarant has prepared initial Use Coidelines and Restrictions which contain general provisions applicable to all of the Properties, as well as specific provisions which may vary within the Properties depending upon the location, characteristics, and intended use. Such initial Use Guidelines and Restrictions are set forth in Section 12.6. Based upon these Use Guidelines and Restrictions, the Board shall adopt the initial rules at its initial organizational meeting.

12.2 <u>Board Power</u>. Subject to the terms of this Article XII and to its duty of care and undivided loyalty to the Association and its Bembers, the Board shall implement and manage the Use Guidelines and Restrictions through rules which adopt, modify, cancel, limit, create exceptions to, or expand the Use Guidelines and Restrictions. Prior to any such action, the Board shall conspicuously publish motice of the proposal at least five business days prior to the Board meeting at which such action is to be considered. Owners

shall care a reasonable opportunity to be heard at a Board meeting prior to action being taken.

The Board shall send a copy of any proposed new rule or amendment to each Owner at least 30 days prior to its effective date.

The rule shall become effective unless disapproved at a meeting by Voting Members representing at least 66 2.3% of the total Class "A" rules and by the Declarant so long as the Declarant owns properly subject to this Declaration, or which may become subject to the Declaration.

The Board shall have all powers necessary and proper, subject to its exercise of sound business judgment and reasonableness, to effect the powers contained in this Section 12.2.

The Board shall provide, without cost, a copy of the Ose Guidelines and Restrictions and rules then in effect to any requesting Rember or Mortgagee.

- 12.3 Members' Power. The Voting Members, at a meeting duly called for such purpose as provided in Section 2.4 of the By-Laws, may adopt, repeal, modify, limit, and expand Use Guidelines and Restrictions and implementing rules by a vote of 66 2/3% of the total Class "A" votes and the approval of the Declarant so long as the Declarant owns property subject to this Declaration or which may become subject to the Declaration.
- 12.4 Owners Acknowledgment. All Owners are subject to the Use Guidelines and Restrictions and are given notice that (a) their ability to use their privately owned property is limited thereby, and (b) the Board may add, delete, modify, create exceptions to, or amend the Use Guidelines and Restrictions in accordance with Sections 12.2, 12.3, and 17.2.

Back Owner by acceptance of a deed acknowledges and agrees that the use and enjoyment and marketability of his or her property can be affected by this provision and that the Use Guidelines and Restrictions and rules may change from time to time.

- 12.5 <u>Rights of Owners.</u> Except as may be specifically set forth in Section 12.6, the Board may not adopt any rule in violation of the following provisions:
 - (a) <u>Equal Treatment</u>. Similarly situated Owners and occupants shall be treated similarly.
- (b) Bousehold Composition. No rule shall interfere with the freedom of occupants of Units to determine the composition of their households, except that the Association shall have the power to require that all occupants be members of a single housekeeping that the total number of occupants permitted in each Unit on the basis of the size and facilities of the Unit and its fair share use of the Common Area, including parking.
- (c) Activities Within Unit. No rule shall interfere with the activities carried on within the confines of Units, eacept that the Association may prohibit activities not normally associated with property restricted to residential use, and it may or safety of occupants of other Units, that generate excessive noise or traffic, that create unsightly conditions visible outside the Unit, that block the views from other Units, or that create an unreasonable source of annoyance.
- (d) Pets. No rule prohibiting the keeping of ordinary bousehold pets shall be adopted over the objection of any affected Owner expressed in writing to the Association. The Association may adopt reasonable regulations designed to minimize damage and disturbance to other Owners and occupants, including regulation requiring damage deposits, waste removal, leash controls, ocise controls, occupancy limits based on size and facilities of the Unit, and fair share use of the Common Area. Nothing in this provision shall prevent the Association from requiring removal of any animal that represents an actual threat to the health or safety of residents or from requiring abatement of any nuisance or unreasonable source of annoyance.
- (e) Allocation of Burdens and Benefits. The initial allocation of financial burdens and rights to use Common Areas among the various Units shall not be changed to the detriment of any Owner over that Owner's objection expressed in writing to the Association. Nothing in this provision shall prevent the Association from changing the Common Areas available, from adopting generally applicable rules for use of Common Areas, or from denying use privileges to those who abuse the Common Area, violate rules of this Declaration, or fail to pay assessments. This provision does not affect the right to increase the amount of assessments as

provided in Article X.

(f) <u>Rights to Develop</u>. We rule or actions by the Association or Board shall unreasonably impede Declarant's right to develop in accordance with the Master Plan.

12.6 Initial Use Guidelines and Restrictions.

- (a) General. The Properties shall be used only for residential, recreational, and related purposes (which may include, without limitation, offices for any property manager retained by the Association or business offices for the Declarant or the Association consistent with this Declaration and any Supplemental Declaration).
- (b) <u>Restricted Activities</u>. The following activities are prohibited within the Properties unless expressly authorized by, and then subject to such conditions as may be imposed by, the Board:
- (i) posting of signs of any kind except those required by law, including posters, circulars and biliboards, provided that the Declarant shall be entitled to post signs consistent with the development concept;
- (ii) parking of commercial rehicles, recreational rehicles, mobile homes, hoats or other watercraft, or other oversized vehicles, stored vehicles or imagerable vehicles in places other than enclosed garages;
- (iii) subdivision of a Unit into two or more Units after a subdivision plan including such Unit has been approved and filed with the appropriate governmental authority, or changing the boundary lines of any Unit, except that the Declarant shail be permitted to subdivide or change the boundary lines of Units which owns;
- (iv) active use of lakes, ponds, streams, or other bodies of water within the Properties or within any Golf Course, except that the owner of the Golf Course and its agents, successors and assigns, shall have the exclusive right and exsement to retrieve golf balls from bodies of water within the Common Areas which are within range of golf balls hit from the Golf Course. The Association shall not be responsible for any loss, damage, or injury to any person or property arising out of the authorized or unauthorized use of lakes, ponds, streams or other bodies of water within or adjacent to the Properties;
- (v) operation of a fraction-sharing program exceeding five Owners whereby the right to exclusive use of the Unit rotates among participants in the program on a fixed or floating time schedule over a period of years, except that Declarant and its assigns may operate such a program with respect to Units which it owns;
- (vi) occupancy of a Unit by more than a single family. For purposes of this restriction, a single family shall be defined as any number of persons related by blood, adoption, or marriage living with not more that one person who is not so related as a single homsehold unit, or no more than three persons who are not so related living together as a single homsehold unit.
- (vii) capturing, trapping or killing wildlife within the properties, except in circumstances posing an imminent threat to the safety of persons using the Properties, and raising, breeding or keeping animals, livestock, or poultry of any kind, except that a reasonable number of dogs, cats, or other usual and common household pets may be permitted in a Unit. Bowever, those pets which are permitted to room free, or, in the sole discretion of the Board, make objectionable noise, endanger the health or safety of, or constitute a nuisance or inconvenience to the occupants of other Units shall be removed upon request of the Board. If the pet owner fails to bonor such request, the Board may remove the pet;
- (viii) activities which materially disturb or destroy the vegetation, wildlife, or air quality within the properties, or which use excessive amounts of water, or which result in unreasonable levels of sound or light poliution;
- (ix) any construction, erection, or placement of ornamentation or other objects or equipment, permanently or temporarily, on the outside portions of the Unit, whether such portion is improved or unimproved;
 - (x) the discharge of firearms within the Properties is prohibited. The term "firearms" includes B-B goms,

pellet guas, bows and arrows, and other firearms of all types, regardless of size. Notwithstanding anything to the contrary contained berein, or in the By-laws, the Association shall not be obligated to take action to enforce this provision; and

(xi) Any business, trade, garage sale, noving sale, runnage sale, or similar activity, except that an Owner or occupant residing in a Unit may conduct business activities within the Unit so long as (a) the existence or operation of the business activity is not apparent or detectable by sight, sound, or small from outside the Unit; (b) the business activity conforms to all roning requirements for the Properties; (c) the business activity does not involve regular visitation of the Unit by clients, customers, suppliers, or other business invites or door-to-door solicitation of residents of the Properties; and (d) the business activity is consistent with the residential character of the Properties and does not constitute a nuisance, or a hazardous or offensive use, or threaten the security or safety of other residents of the Properties, as may be determined in the sole discretion of the Board.

This subsection shall not apply to any activity conducted by the Declarant or a Builder approved by the Declarant with respect to its development and sale of the Properties or its use of any Boits which it owns within the Properties, including the operation of a timeshare or similar program.

The leasing of a Unit shall not be considered a business or trade within the meaning of this subsection. "Leasing", for purposes of this Declaration, is defined as regular, exclusive occupancy of a Unit by any person other than the Owner, for which the Owner receives any consideration or benefit, including but not limited to a fee, service, gratuity, or emolument. Units may be leased only in their entirety. No fraction or portion may be leased. No structure on Unit other than the primary residential dwelling shall be leased or otherwise occupied for residential purposes. There shall be no subleasing of Units or assignment of leases unless prior written approval is obtained from the Board. All leases shall be in writing.

No transient tenants may be accommodated in a Unit and all leases shall be for an initial term of no less than 30 days, except:

(a) with the prior written consent of the Board, or (b) as initially authorized by Declarant in Exhibit "A" or a Supplemental Declaration for Units located within certain areas. The Owners may not amend this provision to prohibit leasing of Units within certain areas authorized by Declarant for rental to transient tenants and for a term less than 30 days until: (a) 75% of the Units within a particular area are owned by Retail Owners, and (b) with approval of 75% of the Class "A" Members, other than the Declarant, within that particular area.

Notice of any lease, together with such additional information as may be required by the Board, shall be given to the Board by the Unit Owner within ten (10) days of execution of the lease. The Owner must make available to the leasee copies of the Declaration, By-Laws, and the roles and regulations. The Board may adopt reasonable rules regulating leasing and subleasing.

- (c) Prohibited Conditions. The following shall be prohibited within the Properties:
- (i) exterior antennas, aerisls, satellite dishes, or other apparatus for the transmission or reception of television, radio, satellite, or other signals of any kind unless completely contained within the dwelling on the Unit so as not to be visible from outside the dwelling or otherwise approved pursuant to Article XI; provided the Declarant and the Association shall have the right, without obligation, to erect or install and maintain such apparatus for the benefit of all or a portion of the Properties;
- (ii) sprinkler or irrigation systems or wells of any type which draw upon water from creeks, streams, ponds, or other ground or surface waters within the Properties; except that Declarant, the Association and the Golf Course shall have the right to collect and divert storm water runoff from streats and other hard surfaces for irrigation and other purposes;
- (iii) hedges, walls, dog runs, amimal pens, or fences of any kind on any Unit except as approved in accordance with Article XI;
- (iv) open garage doors. Garage doors shall remain closed at all times except when entering and exiting the garage; and
 - (v) excessive exterior lighting on any Unit. The Board shall in its sole discretion determine whether any

بالمنطوخ التفاقين والماء ويراوي والمنافعة

exterior lighting is excessive. Exterior lighting shall be screened as set forth in the Design Guidelines.

Article XIII RASEMBNIS

- 13.1 Basements of Encroachment. There shall be reciprocal appurtenant easements of encroachment, and for maintenance and use of any permitted encroachment, between each Unit and any adjacent Common Area and between adjacent Units due to the unintentional placement or settling or shifting of the improvement constructed, reconstructed, or altered thereon (in accordance with the terms of these restrictions) to a distance of not more than three feet, as measured from any point on the common boundary along a line perpendicular to such boundary. However, in no event shall an easement for encroachment exist if such encroachment occurred due to miliful and knowing conduct on the part of, or with the hoowledge and consent of, an Owner, occupant, or the Association.
- 13.2 Basements for Utilities, Btc. There are bereby reserved unto Deciarant, so long as the Declarant owns any property described on Exhibits "A" or "B" of this Declaration, the Association, and the designees of each (which may include, without limitation, Deschutes County, Oregon and any utility) access and maintenance easements upon, across, over, and under all of the Properties to the extent reasonably noressary for the purpose of replacing, repairing, and maintaining cable television systems, master television actemns systems, security and similar systems, roads, walkways, bicycle pathways, trails, ponds, wetlands, irrigation systems, drainage systems, street lights, signage, and all utilities, including, but not limited to, water, sewers, meter boxes, telephone, gas, and electricity, and for the purpose of installing any of the foregoing on property which it owns or within easements designated for such purposes on recorded plats of the Properties. This easement shall not entitle the holders to construct or install any of the foregoing systems, facilities, or utilities over, under or through any existing dwelling on a Unit, and any damage to a Unit resulting from the exercise of this easement shall promptly be repaired by, and at the expense of, the person exercising the easement. The exercise of this easement shall not unreasonably interfere with the use of any Unit and, except in an emergency, entry onto any Unit shall be made only after reasonable notice to the Owner or occupant.

Declarant specifically grants to the local water supplier, electric company, telephone company, cable TV company, and natural gas supplier easements across the Properties for ingress, egress, installation, reading, replacing, repairing, and maintaining utility meters and boxes. However, the exercise of this easement shall not extend to permitting entry into the dwelling on any Unit, nor shall any ctilities be installed or relocated on the Properties, except as approved by the Beclarant.

13.3 Basements to Serve Additional Property. The Declarant bereby reserves for itself and its duly authorized agents, representatives, employees, successors, assigns, licensees, and mortgagees, an easement over the Common Area for the purposes of enjoyment, use, access, and development to the property described in Exhibit "B", whether or not such property is made subject to this Declaration. This easement includes, but is not limited to, a right of ingress and egress over the Common Area for construction of roads and for connecting and installing utilities on such property. Declarant agrees that it and its successors or assigns shall be responsible for any damage caused to the Common Area as a result of vehicular traffic connected with development of such property. Declarant futher agrees that if the easement is exercised for permanent access to such property and such property or any portion thereof is not made subject to this Declaration, the Declarant, its successors or assigns shall enter into a reasonable agreement with the Association to share the cost of maintenance of any access roadway serving such property.

13.4 Rasements for Colf Course.

- (a) Every Unit and the Common Area are burdened with an easement permitting golf balls unintentionally to come upon such Common Area and Units and for golfers at reasonable times and in a reasonable manner to come upon the Common Area or the exterior portions of a Unit to retrieve errant golf balls; provided, however, if any Unit is fenced or walled, the golfer shall seek the Owner's permission before entry. The existence of this essement shall not relieve golfers of liability for damage caused by errant golf balls. Under no circumstances shall any of the following Persons be held liable for any damage or injury resulting from errant golf balls or the exercise of this essement: the Declarant; the Association or its Members (in their capacity as such); IMB Enterprises, its successors, successors—in-title to the Golf Course, or assigns; any successor Declarant; any Builder or contractor (in their capacities as such); any officer, director or partner of any of the foregoing, or any officer or director of any partner.
 - (b) The Owner of the Golf Course, its respective agents, successors and assigns, shall at all times have a right and

- An armanication than signifying

non-erclusive easement of access and use over those portions of the Common Areas reasonably necessary, with or without the use of maintenance, vehicles and equipment, for the operation, maintenance, repair and replacement of the Golf Course,

CARLES TO SERVICE TO S

- (c) The Properties immediately adjacent to the Golf Course are hereby hurdened with a non-exclusive easement in favor of the Golf Course for overspray of mater from any irrigation system serving the Golf Course. Under no circumstances shall the Association or the Owners of the Golf Course be held liable for any damage or injury resulting from such overspray or the exercise of this easement.
- (d) The Owner of the Golf Course, its respective agents, successors and assigns, shall have a perpetual, exclusive easement of access over the properties for the purpose of retrieving golf balls from bodies of water within the Compon Areas lying reasonably within range of golf balls bit from the Golf Course.
- (e) The Owner of the Golf Course, its respective agents, successors and assigns, as well as its guests, invitees, employees, and anthorized users of the Golf Course, shall at all times have a right and non-exclusive easement of access and use over all roadways and golf cart paths, if any, located or to be located within the Properties at reasonable times before, during and after golf tournaments and other similar functions held at the Golf Course.
- (f) The Owner of the Golf Course, its respective agents, successors and assigns, shall have a perpetual non-exclusive easement, to the extent reasonably necessary, over the Properties, for the installation, operation, maintenance, repair, replacement, monitoring and controlling of irrigation systems and equipment, including without limitation wells, pumps and pipelines, serving all or portions of the Golf Course.
- (g) The owner of the Golf Course, its respective agents, successors and assigns, shall have a perpetual, non-exclusive easement to the extent reasonably necessary, over the Properties for the installation, maintenance, repair, replacement and monitoring of atility lines, wires, drainage pipelines and pipelines, serving all or portions of the Golf Course.
- (b) The Properties are hereby burdened with easements in favor of the Golf Gourse for natural drainage of storm water runoff from the Golf Gourse.
- (i) The properties are hereby burdened with easements in favor of the Golf Course for golf cart paths serving the Golf Course. Under no circumstances shall the Association or the Owners of the Golf Course be held liable for any damage or injury resulting from the exercise of this easement.
- (j) The Owner of the Golf Course, its respective agents, successors and assigns, as well as its guests, invitees, employees, and authorized users of the Golf Course, shall at all times have a right and our-exclusive easement to access and use over the golf cart paths located within the Properties.
- 13.5 <u>Basement for Cross-Drainage</u>. Every Unit and the Common Area shall be bordened with easements for natural drainage of storm water runoff from other portions of the Properties, provided no person shall after the natural drainage on any Unit so as to materially increase the drainage of storm water onto adjacent portions of the Properties without the consent of the Owner of the affected property.
- 13.6 Right of Entry. The Association shall have the right, but not the obligation, to enter upon any Unit for emergency, security, and safety reasons, to perform maintenance pursuant to Article V bereof, and to inspect for the purpose of ensuring compliance with this Declaration, any Supplemental Declaration, By-baws, and rules, which right may be exercised by any Bember of the Board, the Association, officers, agents, employees, and managers, and all policemen, firemen, ambulance personnel, and similar emergency personnel in the performance of their duties. Except in an emergency situation, entry shall only be during reasonable hours and after notice to the Owner. This right of entry shall include the right of the Association to enter upon any Unit to cure any conditions which may increase the possibility of a fire or other hazard in the event an Owner fails or refuses to cure the condition within a reasonable time after requested by the Board, but shall not authorize entry into any single family detached dwelling without permission of the Owner, except by emergency personnel acting in their official capacities.
 - 13.7 Maintenance. Some lots in this development are located adjacent to the Golf Course located within the Properties. Owners

shall recognize that the proper maintenance of the Golf Course requires the use of mechanized equipment such as tractors, lawn mowers, fertilizer and/or pesticide applicators, and sprinkler systems, etc.. Some of this equipment must be operated in the early morning hours or late in the evening when the Golf Course is not in use. By acquiring a Unit which is located near the Golf Course, each such Person consents to the use of such equipment in a manner which is customary in the operation of Golf Courses in the Gentral Oregon area.

Proper golf course maintenance requires adequate post management. The golf course operator shall be allowed to take any actions necessary to control posts on the Golf Course or adjacent Properties without any interference.

Article MIY BORTCAGE PROVISIONS

The following provisions are for the benefit of holders, insurers and guarantors of first Mortgages on Units in the Properties. The provisions to this Article apply to both this Declaration and to the By-Laws, notwithstanding any other provisions contained herein.

- 16.1 <u>Motices of Action</u>. An institutional holder, insurer, or guarantor of a first Mortgage who provides written request to the Association (such request to state the name and address of such holder, insurer, or guarantor and the street address of the Unit to which its Mortgage relates, thereby becoming an "Bligible Holder"), will be entitled to timely written notice of:
- (a) any condemnation loss or any casualty loss which affects a material portion of the Properties or which affects any Unit on which there is a first Mortgage held, insured, or goaranteed by such Eligible Holder;
- (b) any delinquency in the payment of assessments or charges owed by a Unit subject to the Mortgage of such Bligible Holder, where such delinquency has continued for a period of 60 days, or any other violation of the Declaration or Ry-Laws relating to such Unit or the Owner or occupant which is not cured within 60 days. Motwithstanding this provision, any holder of a first Mortgage is entitled to written notice upon request from the Association of any default in the performance by an Owner of a Unit of any obligation coder the Declaration or By-Laws which is not cured within 60 days:
 - (c) any lapse, cancellation, or material modification of any insurance policy maintained by the Association; or
 - (d) any proposed action which would require the consent of a specified percentage of Eligible Holders.
- 14.2 Special FULMC Provision. So long as required by the Federal Home Loam Mortgage Corporation, the following provisions apply in addition to and not in lieu of the foregoing. Unless at least 66 2/3% of the first Mortgagees or Members representing at least 66 2/3% of the total Association entitled to vote, consent, the Association shall not:
- (a) by act or omission seek to abandon, partition, subdivide, encumber, sell, or transfer all or any portion of the real property comprising the Common Area which the Association owns, directly or indirectly (the granting of easements for public utilities or other similar purposes consistent with the intended use of the Common Area shall not be deemed a transfer within the meaning of this subsection);
- (b) change the method of determining the obligations, assessments, dues, or other charges which may be levied against an Owner of a Unit;
- (c) by act or omission change, waive, or abandon any scheme of regulation or enforcement pertaining to architectural
 design, exterior appearance or maintenance of Units and the Common Area (the issuance and amendment of architectural standards,
 procedures, rules and regulations, or use restrictions shall not constitute a change, waiver, or abandonment within the meaning of
 this provision);
 - (d) fail to maintain insurance, as required by this Beclaration; or

 (e) use bazard insurance proceeds from any Common Area losses for purposes other than the repair, replacement, or reconstruction of such property.

First Mortgages may, jointly or singly, pay cares or other charges which are in default and which may or have become a charge against the Common Area and may pay overdue premiums on casualty insurance policies or secure new casualty insurance coverage upon the lapse of an Association policy, and first Hortgagees making such payment shall be entitled to immediate reimbursement from the Association.

- 14.3 Other Provision for First Lien Rolders. To the extent possible under Oregon Law:
- (a) any restoration or repair of the Properties after a partial condemnation or damage due to an insurable hazard shall be performed substantially in accordance with this Declaration and the original plans and specification unless approval is obtained from the Eligible Holders of the first Mortgages on Units to which at least SII of the votes of Units subject to Mortgages held by such Bligible Holders are allocated.
- (b) any election to terminate the Association after substantial destruction or a substantial taking in condemnation shall require the approval of the Eligible Holders of first Mortgages on Worlds to which at least 51% of the votes of Units subject to Mortgages held by such Eligible Holders are allocated.
- 14.4 Amendments to Documents. The following provisions do not apply to amendments to the constituent documents or termination of the Association made as a result of destruction, damage, or condemnation pursuant to Section 14.3(a) and (b), or to the addition of land in accordance with Article IX.
- (a) The consent of Voting Members representing at least 66 2/3% of the Class "A" votes and of the Declarant, so long as it owns any land subject to this Declaration, and the approval of the eligible holders of first Mortgages on Units to which at least 66 2/3% of the votes of Units subject to a Mortgage appertain, shall be required to terminate the Association.
- (b) The consent of Voting Members representing at least 66 2/3% of the Class "A" notes and of the Declarant, so long as it owns any land subject to this Declaration, and the approval of Bligible Bolders of first Mortgages on Units to which at least 51% of the votes of Units subject to a Mortgage appertain, shall be required to materially amend any provisions of the Declaration, By-Laws, or Articles of Interporation, or to add any material provisions thereto which establish, provide for, govern, or regulate any of the following:
 - (i) votine:
 - (ii) assessments, assessment liens, or subordination of such liens;
 - (iii) reserves for maintenance, repair, and replacement of the Common Area;
 - (iv) insurance or fidelity bonds;
 - (v) rights to use the Common Area;
 - (vi) responsibility for maintenance and repair of the Properties;
 - (vii) expansion or contraction of the Properties or the addition, annexation, or withdrawal of Properties to or from the Association:
 - (viii) boundaries of any Unit;
 - (ix) leasing of Units;
 - (r) imposition of any right of first refusal or similar restriction of the right of any Owner to sell, transfer, or otherwise convey his or her Unit:
 - (xi) establishment of self-management by the Association where professional management has been required by an Eligible Holder; or
 - (xii) any provisions included in the Declaration, By-Laws, or Articles of Incorporation which are for the express benefit of bolders, guarantors, or insurers of first Mortgages on Units.
- 14.5 No Priority. No provision of this Declaration or the By-Laws gives or shall be construed as giving any Owner or other party priority over any rights of the first Mortgagee of any Unit in the case of distribution to such Gwner of insurance proceeds or condemnation awards for losses to, or a taking of, the Common Area.

- 14.6 <u>Notice to Association</u>. Opon request, each Owner shall be obligated to furnish to the Association the name and address of the holder of any Hortgage encumpering such Owner's Unit.
- 14.7 <u>Amendment by Sgard</u>. Should the Federal Rational Mortgage Association or the Federal Rome Loan Mortgage Corporation subsequently delete any of its respective requirements which necessitate the provisions of this Article or make any such requirements less stringent, the Board, without approval of the Owners, may record an amendment to this Article to reflect such changes.
- 14.8 Applicability of Article XIV. Mothing contained in this Article shall be construed to reduce the percentage rote that must otherwise be obtained under the Declaration, By-Laws, or Oregon law for any of the acts set out in this Article.
- 14.9 Pailore of Bortgagee to Respond. Any Mortgagee who receives a written request from the Roard to respond to, or consent to, any action shall be deemed to have approved such action if the Association does not receive a written response from the Mortgagee within thirty (30) days of the date of the Association request, provided such request is delivered to the Mortgagee by certified or registered mail, return receipt requested.

Article XV DECLARAST'S RICETS

15.1 Transfer. Any or all of the special rights and obligations of the Declarant set forth in this Declaration or the By-Laws may be transferred to other Persons, provided that the transfer shall not reduce an obligation nor enlarge a right beyond that contained in this Declaration or the By-Laws. We such transfer shall be effective nuless it is in a written instrument signed by the Declarant and duly recorded in the Office of the County Clerk. Mothing in this Declaration shall be construed to require Declarant or any successor to develop any of the property set forth in Exhibit "B" in any manner whatsoever.

So long as construction and initial sales of Units shall continue, the Declarant and Builders authorized by Declarant may maintain and carry on upon portions of the Common Area such facilities and activities as, in the sole opinion of the Declarant, may be reasonably required, convenient, or incidental to the construction or sale of such Units, including but not limited to business offices, signs, model units, and sales offices. The Declarant's or Builder's unilateral right to use the Common Area for the purposes stated in this paragraph shall not be exclusive and shall not unreasonably interfere with use of such Common Areas by Owners unless leased pursuant to a lease agreement with the Association providing for payment of reasonable rent.

No Person shall record any declaration of covenants, conditions and restrictions, or declaration of condominium or similar instrument affecting any portion of the Properties without Declarant's review and written consent. Any attempted recordation without such consent shall result in such instrument being void and of no force and effect unless subsequently approved by recorded consent signed by the Declarant.

The Beclarant, in its sole discretion, may convert all or part of the Properties in Bubbbits "A" and/or "B" to a "mini destination resort" or "destination resort" ("Resort") as those terms are now or hereafter defined in the Deschutes County Zoning Ordinance or any other applicable ordinance. If additional or substitute restrictions are appropriate or required under the Resort designation, the Declarant shall have sole discretion in making those changes. By purchasing a Unit in ASPEN LAKES each Owner expressly consents to such conversion to Resort status and to such additional or substitute restrictions and no further consent shall be required from any Owner. Bowever, these additional or substitute restrictions shall not apply to those Units in Section 1, Township 15 South, Range 10 East, of the Willamette Meridian, except as may be necessary to establish Meighborhoods as described in Section 1.23 of this Declaration.

This Article may not be amended without the written consent of the Declarant. The rights contained in this Article shall terminate upon the earlier of (a) 40 years after the conveyance of a Unit to a Retail Owner, or (b) upon recording by Declarant of a written statement that all sales activity has ceased. Thereafter, the Declarant or Builders may continue to use the Common Areas for purposes stated in this Section only pursuant to a rental or lease agreement between the Declarant or such Builder and the Association which provides for rental payments based on the fair market rental value of any such portion of the Common Areas.

- 15.2 Ownership and Operation of Golf Course. All Persons, including all Owners, are hereby advised that no representations or warrancies have been or are made by the Declarant or any other Person with regard to the continuing existence, ownership or operation of the Golf Course, if any, and no purported representation or warranty in such regard, either written or oral, shall ever be effective without an amendment to this Declaration executed or joined into by the Declarant. Further, the ownership and/or operation of the Golf Course, if any, may change at any time and from time to time by virtue of, but without limitation, (a) the ownership and/or operations of the Golf Course by an independent entity or entities; (b) the creation or conversion of the ownership and/or operating structure of the Golf Course to an "equity" club or similar arrangement whereby the Golf Course or the rights to operate it are transferred to an entity which is owned or controlled by its members; or (c) the transfer of ownership or control of the Golf Course to one or more affiliates, shareholders, employees, or independent contractors of the Declarant. No consent of the Association or any Owner shall be required to effectuate such transfer or conversion.
- 15.3 Right to Use. Meither membership in the Association nor Ownership or occupancy of a Unit shall confer any ownership interest in or right to use the Golf Course. Rights to use the Golf Course will be granted only to such persons, and on such terms right from time to time in its sole and absolute discretion and without notice, to amend or waive the terms and conditions of use of the Golf Course, including without limitation eligibility for and duration of use rights, categories of use and extent of use privileges, and number of users, and shall also have the right to reserve use rights and to terminate use rights altogether, subject to the provisions of any outstanding membership documents.
- 15.4 Yiew Impairment. Meither the Declarant, the Association nor the owner or operator of the Golf Course guarantees or represents that any view over and across the Golf Course from adjacent Units will be preserved without impairment. The owner of the Golf Course, if any, shall have no obligation to prune or this trees or other landscaping, and shall have the right, in its sole and absolute discretion, to add trees and other landscaping to the Golf Course from time to time. In addition, the owner of the Golf Course may, in its sole and absolute discretion, change the location, configuration, size and elevation of the tees, bunkers, fairways and greens on the Golf Course from time to time. Any such additions or changes to the Golf Course may diminish or obstruct any view from the Units and any express or implied easements for rice purposes or for the passage of light and air are hereby expressly disclaimed.

Declarant may choose to, but is not required to, construct a Club Bouse facility to service the Golf Course. To enable that facility to enjoy a view of the Cascade Mountain Range, it may be necessary to remove or prume certain trees and regetation on Common Area, Units numbered 1 through 3, 19 through 23, and 35 through 63 on plat no. 90-727 "Golf Course Astates at Aspen Lakes" Deschutes County, Oregon. Builders or Owners of said lots hereby agree to the removal of or pruning of such trees and vegetation on said lots as may be necessary for operator of said Golf Course to obtain and preserve views of the Cascade Range from Club House facility.

- 15.5 Limitations on Amendments. In recognition of the fact that the provisions of this Article are for the benefit of the owner of the Golf Course, no amendment to this Article, and no amendment in derogation of any rights reserved or granted to the owner of the Golf Course by other provisions of this Declaration, may be made without the written approval of the owner of the Golf Course. The foregoing shall not apply, however, to amendments made by the Declarant.
- 15.6 <u>Jurisdiction and Gooperation</u>. It is the Declarant's intention that the Association and the owner of the Golf Course shall cooperate to the maximum extent possible in the operation of the Properties and the Golf Course. Each shall reasonably assist the other in upholding the Community-Nide Standard. The Association shall have no power to promulgate rules and regulations affecting activities on or use of the Golf Course.

Article XVI DISPUTE RESOLUTION AND LINITATION ON LITIGATION

16.1 Agreement to Avoid Costs of Litigation and to Limit Right to Litigate Disputes. The Association, Declarant, Owners, all Persons subject to this Declaration, and any Person not otherwise subject to this Declaration who agrees to submit to this Article (collectively, "Bound Parties"), agree to encourage the amicable resolution of disputes involving the Properties, and to avoid the emotional and financial costs of litigation if at all possible. Accordingly, each Bound Party corenants and agrees that all claims, grievances or disputes between such Bound Party and any other Bound Party involving the Properties, including without limitation

claims, grievances or disputes priving out of or relating to the interpretation, application or enforcement of this Declaration, the By Lams, the Association rules, or the Articles of Incorporation (collectively "Claim"), except for those claims authorized in Section 16.2, shall be subject to the procedures set forth in Section 16.3.

- 16.2 Exempt Glaims. The following Claims ("Exempt Glaims") shall be exempt from the provisions of Section 16.3:
 - (a) any suit by the Association against any Bound Party to enforce the provisions of Article X;
- (b) and suit by the Association to obtain a temporary restraining order (or equivalent emergency equitable relief) and such other ascillary relief as the court may deem necessary in order to maintain the status quo and preserve the Association's ability to enforce the provisions of Article XI and Article XII; and
- (a) any suit between Owners (other than Declarant) seeking redress on the basis of a Claim which would constitute a course of artion under the law of the State of Oregon in the absence of a Claim based on the Declaration, By-Laws, Articles or rules of the Association, if the amount in controversy exceeds \$5,000.00.

Any Bound Party having an Exempt Claim may submit it to the alternative dispute resolution procedures set forth in Section 16.3, but there shall be no obligation to do so.

- 16.3 Mandatory Procedures For All Other Claims. Any Bound Party having a Claim ("Claimant") against any other Bound Party ("Respondent"), other than a Claim exempted from this provision by Section 16.2, shall not file suit in any court or initiate any proceeding before any administrative tribunal seeking redress or resolution of such Claim nutil it has complied with the following
- (a) Notice. The Claimant shall notify each Respondent in writing of the Claim ("Notice"), stating plainly and concisely:
 - (i) the nature of the Claim, including date, time, location, persons involved, Respondents's role in the Giaim and the provisions of this Declaration, the By-Laws, the rules, the Articles of Incorporation or other authority out of which the Claim arises;

(ii) the basis of the Claim (i.e., the provisions of the Declaration, By-Laws, rules or Articles triggered by the Claim);

(iii) what Claimant mants Respondent to do or not do to resolve the Claim; and

(ir) that Claimant wishes to resolve the Glaim by mutual agreement with Respondent, and is willing to meet in person with Respondent at a mutually agreeable time and place to discuss in good faith ways to resolve the Clain.

(b) Regotiation.

(i) Each Claimant and Respondent ("Parties") shall make every reasonable effort to meet in person and confer for the purpose of resolving the Glaim by good faith negotiation.

(ii) Upon receipt of a written request from any Party, accompanied by a copy of the Motice, the Board may appoint a representative to assist the Parties in resolving the dispute by negotiation, if in its discretion it believes its efforts will be beneficial to the Parties and to the welfare of the community.

(c) Mediation.

(i) If the Parties do not resolve the Claim through negotiation within 30 days of the date of the Notice (or within such other period as may be agreed upon by the Parties) ("Termination of Megotiations"), Claimant shall have 30 additional days within which to submit the Claim to mediation under the auspices of the deighborhood dediation Genter in Portland,Oregon, or such other independent agency providing similar services upon which the parties may mutually agree.

(ii) If Claimant does not submit the Claim to mediation within 30 days after Termination of Degotiations, Claimant shall be deemed to have waived the Claim, and Respondent shall be released and discharged from any and all liability to Claimant on account of such Ulaim; provided, withing hardin shall release or discharge Respondent from any liability to Persons not a Party to the foregoing proceedings.

(iii) If the Parties do not settle the Claim within 30 days after submission of the matter to the mediation process, the mediator shall, within five days of the termination of the mediation proceedings, provide the Farties with a written non-binding recommendation for recolution of the Claim ("Mediator's Recommendation").

(d) Final and Binding Arbitration.

(i) If the Parties do not agree in writing to accept the Mediator's Recommendation within 10 days after receipt of notice thereof, the Ciaimant shall have 30 additional days to submit the Claim to arbitration in accordance with the Rules of Arbitration contained in Exhibit "D" or the Ciaim shall be deemed abandoned, and Respondent shall be released and discharged from any and all liability to Claimant arising out of such Claim; provided nothing herein shall release or discharge Respondent from any liability to Persons not a Party to the foregoing proceedings.

(ii) This subsection (d) is an agreement of the Bound Parties to arbitrate all Claims except Brempt Claims and is specifically enforceable under the applicable arbitration law of the State of Oregon. The arbitration award (the "award") shall be final and binding, and judgment may be entered upon it in any court of competent

jurisdiction to the fullest extent permitted under the laws of the State of Oregon,

16.4 Allocation of Costs of Resolving Claims.

- (a) Each Party shall bear all of its own costs incurred prior to and during the proceedings described in Section 16.3 (a), (b), and (c), including the fees of its attorney or other representative. Back Party shall share equally all charges rendered by the mediator(s) pursuant to Section 16.3(c).
- (b) Each Party shall bear all of its own costs (including the fees of its attorney or other representative) incurred after the termination of mediation under Section 16.3(e) and shall share equally in the costs of conducting the arbitration proceeding (collectively, "Post Mediation Costs"), except as otherwise provided in this subsection; provided, however, if the Claim is rejected in whole or in part, the Claimant shall pay all Post Mediation costs, including the costs incurred by the Respondent.
- 16.5 Enforcement of Resolution. If the Parties agree to resolve any Claim through negotiation or mediation in accordance with Section 16.3 and any Party thereafter fails to abide by the terms of such agreement, or if the Parties agree to accept the Award following arbitration and any Party thereafter fails to comply with such Award, then any other party may file suit or initiate administrative proceedings to enforce such agreement of Award without the need to again comply with the procedures set forth in Section 16.3. In such event, the Party taking action to enforce the agreement or Award shall be entitled to recover from the non-complying Party (or if more than one non-complying Party, from all such Parties pro rata) all costs incorred in enforcing such agreement or Award, including without limitation attorney fees and court costs.

Article XVII GENERAL PROVISIONS

17.1 Term. This Declaration shall run with and bind the Properties, and shall incre to the benefit of, and shall be enforceable by the Association or any Owner, their respective legal representatives, heirs, successors, and assigns, for a term of 40 years from the date this Declaration is recorded. After such time, this Declaration shall be automatically extended for successive periods of 10 years, unless an instrument in writing, signed by a majority of the the Owners, has been recorded within the year preceding each extension, agreeing to amend, in whole or in part, or terminate this Declaration, in which case this Declaration shall be amended or terminated as specified therein.

17.2 Amendment.

(a) Dr Declarant.

- (1) Prior to the conveyance of the first Unit to a Retail Owner, Declarant may unilaterally amend this Declaration. After the conveyance of any Unit, the Declarant may unilaterally amend this Declaration if such amendment is;
 - (i) decessary to bring any provision into compliance with any applicable governmental statute,

(ii) necessary to enable any reputable title insurance company to issue title insurance coverage on the Waits;

- (iii) required by an institutional or governmental lender, purchaser, insurer or guarantor of Mortgage loans, including for example the Federal Mational Mortgage Association or Federal Home Loan Mortgage Corporation, to enable it to make, purchase, insure or guarantee Mortgage loans on the Units; or
- (iv) otherwise necessary to satisfy the requirements of any governmental agency for approval of this Declaration. This Declaration may be amended in accordance with Section 17.2(b).
- (2) Declarant may at any time amend this Declaration without consent of Owners, Board, or Association in connection with any conversion of the property to a Destination Resort as allowed in Section 15.1 hereis.
- (b) 87 Owners. Unless the Declarant has the right to amend this Declaration in accordance with Section 17.2(a), this Declaration may be amended only by the affirmative vote or written consent, or any combination thereof, of Yoting Members representing by the vote or written consent of the Yoting members representing at least (a) 75% of the Members; and (b) 75% of the Members there other than the Declarant.

In addition, the approval requirements set forth in Article XIV hereof shall be met if applicable. Motwithstanding the above, the percentage of votes necessary to amend a specific clause shall not be less than the prescribed percentage of affirmative votes required for action to be taken upder that clause.

(c) <u>Validity and Effective Date of Amendments</u>. Amendments to this Declaration shall become effective upon their being recorded in the Office of the County Clerk unless a later effective date is specified therein. Any procedural challenge to an amendment must be made within one year of its recording or such amendment shall be presumed to have been validly adopted. In no event shall a change of conditions or circumstances operate to amend any provisions of this Declaration.

If an Owner consents to any amendment to this Declaration or the By-Laws, it will be conclusively presumed that such Owner has the authority to so consent, and no contrary provision in any Mortgage or contract between the Owner and a third party will affect the validity of such amendment.

No amendment may remove, revoke, or modify any right or privilege of the Declarant without the written consent of the Declarant or the assignee of such right or privilege.

- 17.3 Severability. Invalidation of any provision of this Declaration, in whole or in part, or any application of a provision of this Declaration by indepent or court order shall in no way affect other provisions or applications.
- 17.4 <u>Perpetuities.</u> If any of the covenants, conditions, restrictions, or other provisions of the Declaration shall be unlawful, you'd, or voidable for violation of the rule against perpetuities, then such provisions shall continue only until twenty-one years after the death of the last survivor of the new living descendants of Elizabeth II, Queen of England.
- 17.5 <u>Litigation</u>. No judicial or administrative proceeding shall be consensed or prosecuted by the Association unless approved by Yoting Members representing at least a majority of a quorum of the Class "A" Members. This Section shall not apply, however, to (a) actions brought by the Association to enforce the provisions of this Declaration (including, without limitation, the foreclosure

- of liens); (b) the impositions and collection of assessments as provided in Article X; (c) proceedings involving challenges to advantage transition; or (d) counterclaims brought by the Association in proceedings instituted against it. This Section shall not be amended unless such amendment is approved by the percentage of votes, and pursuant to the same procedures, necessary to institute proceedings as provided above. This provision shall apply in addition to the provisions of Article XVI, if applicable.
- 17.6 Use of the Words "ASPEN LAKES". We Person shall use the words "ASPEN LAKES" or any derivative, or any other term which Declarant may select as the name of this development or any component thereof, in any printed or promotional material without the Declarant's prior written consent. However, Owners may use the words "ASPEN LAKES" in printed or promotional matter solely to specify that particular property is located within the Properties and the Association shall be entitled to use the words "ASPEN LAKES" in its name.
- 17.7 Compliance. Every Owner and occupant of any Unit shall comply with this Declaration, the By-baws, and the rules of the Association. Failure to comply shall be grounds for an action to recover sums due, for damages or injunctive relief, or for any other remedy available at law or in equity, by the Association or, in a proper case, by an aggrieved Unit Owner(s).
- 17.8 <a href="https://doi.org/10.1001/j.com/html
- 17.3 Enforcement of Bonded Obligations. In the event improvements to the Common Area have not been completed as may be required according to approval provisions covering the Properties by Deschutes County or other government entity authorizing the commencement of Unit sales (Commencement of Sales) and the Association, as a result of such lack of completion, is an obligee under a bond or other arrangement ("Bond") to secure performance of the commitment of the Declarant or a Builder to complete such improvements, the following shall apply:
- (a) The Board shall consider and vote upon the question of action by the Association to enforce the obligations under the Boad with respect to any improvements for which a "Motice of Completion" has not been filed within 60 days after the completion date specified for such improvement in the "Planued Construction Statement" appended to the Bond. If the Association has given an extension in writing for the completion of any Common Area improvement, the Board shall consider and vote on the aforesaid question if a "Motice of Completion" has not been filed within 30 days after the expiration of such an extension.
- (b) In the event the Board determines not to initiate action to enforce the obligation under the Board or in the event the Board fails to consider and vote upon such question as provided above, the Board shall call a special meeting of the Owners for the purpose of voting to override such decision or such failure to act by the Board. Such meeting shall be called according to the provisions of the By-Laws dealing with special meetings, but in any event, such meeting shall be held not less than 35 nor more than 45 days after receipt by the Board of a petition for such meeting, signed by Voting Members representing not less than 5% of the Bembers of the Association.
- (c) Yoting Members, other than the Declarant, shall be entitled to vote at such a meeting. A vote at such meeting of a majority of the Yoting Members, other than Declarant, to take action to enforce the obligations under the Bond shall be deemed to be the decision of the Association, and the Board shall thereafter implement such decision by initiating and pursuing appropriate action in the name of the Association.

87: Marthur Co
STATE OF OREGON, County of Deschutes, ss: On 3 6, 1993, the undersigned, a Rotary Public in and for said County and State, personally appeared known to me to be a partner of the partnership that executed the within instrument and acknowledged to me that such partnership executed the same. OFFICIAL SEAL CARRIL LYNN DAVIS NCTARY PUBLIC-OREGON NCTARY PUBLIC-OREGON COMMISSION NO. 012851 My COMMISSION EXPIRES JAN. 30. 1996 My COMMISSION EXPIRES JAN. 30. 1996
STATE OF ORIGON, Country of Deschutes, ss: On 3 8 , 1993, the undersigned, a Motary Public in and for said Country and State, personally appeared modification of the partnership that executed the within instrument and acknowledged to me that such partnership executed the same. OFFICIAL SEAL CARGILLYAN GAVIS NOTARY PUBLIC-OREGON NOTARY PUBLIC-OREGON NOTARY PUBLIC-OREGON NOTARY PUBLIC-OREGON NOTARY PUBLIC-OREGON My Commission Expires 130 1906
STATE OF ORECOM, County of Deschutes, ss: On3 S

IN WITNESS WEERBOY, the undersigned Declarant has executed this Amended Declaration this 8 day of March, 1993.

IMB EMTERPRISES, an Oregon Partnership 17204 Bwy 126 Sisters, OR 97759

EXHIBIT "A"

Lots One (1) through Porty-three (43), rights of way for private roads designated as Brian Drive, Joseph Court, Matthew Drive, and Panela Drive, inclusive of GOLF COURSE ESTATES AT ASPEN LAKES, Deschutes County, Oregon.

EXHIBIT "B"

All of the following described property, or any division of, or new mapping thereof:

(1) Tracts A and B, inclusive of GOLP COURSE ESTATES AT ASPEN LAKES, Deschutes County, Oregon;

(2) Tracts A and B, inclusive of TRE RIM AT ASPEN LAKES, Deschutes County, Oregon;
(3) Lots One (1) through Ten (10) in Block Seven (7), Lots One (1) through Seven (7) in Block Sight (8), Lots One (1) through Fourteen (14) in Block Bine (9), inclusive of WILD BORSE MEADOWS, Deschutes Gounty, Oregon;

(4) any real property now owned or hereafter acquired by Declarant within One (1) mile of the perimeter boundary of the above described property. Such property, if not now owned, must be acquired within Ten (10) years of the date of recording of these

EXHIBIT "C"

BY-LAWS

OF

ASPEN LAKES ESTATE OWNERS, INC.

TABLE OF CONTENTS

weight five.	***************************************
I.1	Rane.
1.2	Principal Office
1.3	Principal Office
	Definitions
Article II	• • • • • • • • • • • • • • • • • • • •
2.1	Nowhership
2.2	distribution and the second se
2.3	rade of destings
2.4	worker deerfagetaeretteritteritaereteri
2.5	aberrar decriffs
	motice of meetidage
2.6	
2.7	and not make to the describes
2.8	. A. r
2.9	**************************************
2.10	
2.11	American (44) (44) (40) (40) (41) (41) (41) (41) (41) (41)
2.12	SOURCE OF DECITORS
2.13	Action Without a Heeting
Article III	Contradius Pader Comments
3.1	Governing Body; Composition.
3.2	
3.3	Directors Breing Clane 31 Control B. 1.2
3,4	Directors During Glass "B" Control Period.
3.5	
3.6	processed and sets of Assistances
3.7	Account of Affectors and (Accounted)
3.8	ALBERTAGLIONEL DECCINES.
3.9	wegnret meerings
3.10	sherist meerings
3.11	RATIFE OF MOCICE ***********************************
3.12 3.12	VOLUM OF BURLD OF BILESCOPE
3.13	COURSE OF MEETINGS
3.14	open accounts.
3.15	weeton kirmont a totaal section.
3.16	
3.17	AGCTES44144044444444444444444444444444444444
3.18	aten of class a compet to distribute actions
3.19	Canda Scare Me
3.20	ACCOUNTS AND MEDOL(2)
3.21	DOLLARINGTON
3. 2 2	PIGHUD OF THE VERGETALION
3.23	Raforcement.
3.24	Prohibited Acts
rticle IV	MFG
4.1	ATTIGEDIAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA
4.2	Blection and Term of Office
4.3	Removal and Facaucies
	11

9.9	FOWERS AND DATIES	11
4.5	Resignation	
4.6	Agreements, Contracts, Seeds, Leases, Charle Res	
4.7	Compensation.	12
Article V		
5.1	General.	12
5.2	Covenants Committee.	12
5.3	Transitional Advisory Committee	12
***	Prometrador Housearly Anguiteses the state of the state o	12
Article M		
6.1	Fiscal Tear	12
6.2	Parliamentary Rules	12
6.3	Conflicts	12
6.4	Books and Records.	12
	(a) inspection by Members and Mortespasse.	12
		12
	fol mann on ymphorizonistatistatististististististististististististist	13
6.5	(c) Inspection by Directors.	13
	Hotices	13
6.6	Amendments	12

BY-LAWS

OF

ASPEN LAKES ESTATE OWNERS, INC.

ARTICLE I Mane, Principal Office, and Definitions

- 1.1 Bame. The name of the Association shall be Aspen Lakes Batate Owners, Inc. ("Association").
- 1.2 Principal Office. The principal office of the Association in the State of Oregon shall be located in Deschutes County, Oregon. The Association may have such other offices, either within or conside the State of Oregon, as the Board may determine or as the affairs of the Association may require.
- 1.3 <u>Definitions.</u> The words used in these By-laws shall be given their normal, commonly understood definitions. Capitalized terms shall have the same meaning as set forth in the Declaration of Covenants, Conditions, and Restrictions for Aspen Lakes filed in the Office of the Country Clerk ("Declaration"), unless the context indicates otherwise.

APTICLE II Association: Hembership, Heetings, Oporus, Voting, Proxies

- 2.1 <u>Membership</u>. The Association shall have two classes of membership, Class "A" and Class "B", as more fully set forth in the Declaration, the terms of which pertaining to membership are incorporated by reference.
- 2.2 Place of Meetings. Heetings of the Association shall be held at the principal office of the Association or at such other suitable place convenient to the Members as may be designated by the Board either within the Properties or as convenient thereto as possible and practical.
- 2.3 Annual Beetings. The first meeting of the Association, whether a regular or special meeting, shall be held within 45 days after the conveyance of 51% of the Boits in the Properties to Retail Owners, but not later than six months after the first conveyance of a Unit to a Retail Owner after Commencement of Sales. Subsequent regular annual meetings shall be set by the Board so as to occur at least 30 days but not more than 120 days before the close of the Association's fiscal year on a date and at a time set by the Board.
- 2.4 Special Meetings. The President may call special meetings. In addition, it shall be the duty of the President to call a special meeting if so directed by resolution of the Board or upon a petition signed by foting Members representing at least 10% of the Members of the Association. In accordance with O.R.S. 94.616, the Association shall call a special meeting not later than 120 days after the Class "B" membership has terminated for the purpose of turning over administrative responsibility to the Association. If the Declarant fails to call the meeting, the Temporary Advisory Committee, if any, or any Owner may call the meeting is accordance with O.R.S. 94.609.
- 2.5 <u>Notice of Meetings.</u> Written or printed notice stating the place, day, and hour of any meeting of the Voting Members shall be delivered, either personally or by mail, to each Yoting Member entitled to vote at such meeting, not less than 10 nor more than 50 days before the date of such meeting, by or at the direction of the President, or the Secretary, or the officers, or persons calling the meeting.

In the case of a special meeting or when otherwise required by statute or these By-Laws, the purpose or purposes for which the meeting is called shall be stated in the notice. So business shall be transacted at a special meeting except as stated in the notice.

If mailed, the notice of a meeting shall be deemed to be delivered when deposited in the United States mail addressed to the

The second secon

Yoting Member at his or her address as it appears on the records of the Association, with postage prepaid.

- 2.5 <u>Majver of Motice</u>. Majver of notice of a meeting of the Voting Members shall be deemed the equivalent of proper notice. Any Voting Member may, in writing, waive notice of any meeting of the Members, either before or after such meeting. Attendance at a meeting by a Voting Member shall be deemed waiver by such Voting Member of notice of the time, date, and place thereof, unless such Voting Member specifically objects to lack of proper notice at the time the meeting is called to order. Attendance at a special meeting also shall be deemed waiver of notice of all business transacted unless an objection on the basis of lack of proper notice is raised before the business is put to a vote.
- 2.7 Adjournment of Meetings. If any meeting of the Association cannot be held because a quorum is not present, a majority of the Yoting Members who are present at such meeting may adjourn the meeting to a time and less than 5 nor more than 30 days from the time the original meeting was called. At the reconvened meeting, if a quorum is present, any business may be transacted which might have been transacted at the meeting originally called. If a time and place of reconvening the meeting is not fixed by those in attendance at the original meeting or if for any reason a new date is fixed for reconvening the meeting after adjournment, notice of the time and place for reconvening the meeting after adjournment, notice

The Yoting Members present at a duly called or held meeting at which a quorum is present may continue to do business until adjournment, notwithstanding the withdrawal of enough Yoting Members to leave less than a quorum, provided that Yoting Members representing at least 25% of the Members of the Association remain in attendance, and provided that any action taken is approved by at least a majority of the votes required to constitute a quorum.

- 2.8 <u>Voting.</u> The voting rights of the Members shall be as set forth in the Declaration, and such voting rights provisions are specifically incorporated by reference. Voting of the Voting Members at a meeting may be by voice or ballot, except the election of directors which shall be by secret written ballot.
- 2.9 Proxies. Yoting Hembers may rote by proxy. Ho proxy shall be valid unless it meets the requirement of 0.2.8. 94.660 and is signed by the Owner or his or her duly authorized attorney in-fact, dated, and filed with the Secretary of the Association prior to any meeting for which it is to be effective. Ho proxy shall be valid after one year from its date of execution neless otherwise specified in the proxy.
- 2.10 Majority. As used in these By-Laws, the term "majority" shall mean those votes, owners, or other group as the context may indicate totaling more than 50% of the total eligible number whether represented in person or by proxy.
- 2.11 Quorum. Except as otherwise provided in these By-Laws or in the Declaration, the presence of the Voting Members representing a majority of the total votes in the Association shall constitute a quorum at all meetings of the Association membership, whether personally present or present by proxy.
- 2.12 Conduct of Beetings. The President shall preside over all meetings of the Association, and the Secretary shall keep the minutes of the meeting and record in a minute book all resolutions adopted at the meeting, as well as a record of all transactions occurring at the meeting.
- 2.13 Action Without a Meeting. Any action required or permitted by law to be taken at a meeting of the membership may be taken without a meeting, without prior notice and without a vote if written consent specifically authorizing the proposed action is signed by Voting Members holding at least the minimum number of votes necessary to authorize such action at a meeting if all Voting Members entitled to vote thereon were present. Such consents shall be signed within 60 days after receipt of the earliest dated consent, dated and delivered to the Association at its principal place of business in the State of Oregon. Such consents shall be filed with the minutes of the Association, and shall have the same force and effect as a unanimous vote of the Voting Members.

Board of Directors: Humber, Powers, Sections

A. Composition and Selection.

- 3.1 Governing Body; Composition. The affairs of the Association shall be governed by a Board, each of whom shall have one equal rote. Except with respect to directors (a) appointed by the Class "B" Hember during the Class "B" Control Period pursuant to Section 3.3, (b) elected by the Class "B" Member after the expiration of the Class "B" control Period or (c) elected by the Class "A" votes held by Declarant after the conversion of the Class "B" membership to Class "A" membership pursuant to Section 3.3(b) of the Declaration, the directors shall be Members or spouses of such Members; provided, however, no person and his or her spouse may serve officer of such Member shall be eligible to serve as a director unless otherwise specified by written notice to the Association especies by such Sember. No Member during the Class "B" membership to Class "B" Member during the Class "B" Control Period pursuant to Section 3.3, (b) elected by the Class "B" Member during the Class "B" Control Period, or (c) elected by the Class "A" votes held by Declarant after the conversion of the Class "B" membership to Class "A" membership pursuant to Section 3.3(b) of the Declaration.
- 3.2 <u>Mumber of Directors</u>. The number of directors in the Association shall be not less than three nor more than seven, as provided in Section 3.5. The initial Board shall consist of three directors as identified in the Articles of Incorporation.
- 3.3 <u>Directors During Glass "B" Control Period.</u> Subject to the provisions of Section 3.5, the directors shall be appointed by the Class "B" Member acting in its sole discretion and shall serve at the pleasure of the Glass "B" Member until the first to occur of the following:
 - (a) when 75% of the Maximum Units have certificates of occupancy issued thereon and have been conveyed to Retail Owners;
- (b) Fifteen years from the anniversary date of the conveyance of the first Unit to a Retail Owner after the Commencement of Sales, except as provided for in Section 3.3(b)(ii) of the Declaration or;
 - (c) when, in its discretion, the Class "8" Hember so determines.
- 3.4 <u>Momination of Directors.</u> Except with respect to directors selected by the Class "B" Member, or the Declarant, as the case may be, pursuant to Section 3.1, nominations for election to the Board shall be made by a Sominating Committee. The Sominating Committee shall consist of a Chairman, who shall be a member of the Hoard, and three or more Hembers or representatives of Hembers. The Sominating Committee shall be appointed by the Board not less than 30 days prior to each annual meeting of the Voting Members to serve a term of one year or until their successors are appointed, and such appointment shall be announced at each such annual meeting. The Hominating Committee shall make as many nominations for election to the Board as it shall in its discretion determine, but in no event less than the number of positions to be filled from each slate as provided in Section 3.5. The Hominating Committee thail nominate directors to be elected by the Yoting Hembers. Mominations shall also have a reasonable opportunity to communicate their qualifications to the Yoting Members and to solicit votes.
 - 3.5 Election and Term of Office. Motwithstanding any other provision of these My-Laws
- (a) Within one year after the first Unit is conveyed to a Retail Owner, the President shall call a special meeting to be held at which the Voting Members representing the Class "A" Members, other than the Declarant, shall elect one of the three directors. The Remaining two directors shall be appointed to the Class "B" Member. The director elected by the Voting Members shall not be subject to removal by the Class "B" member and shall be elected for a term of two years or until the happening of the event described in subsection (b) below, whichever is shorter. If such director's term expires prior to the happening of the event described in subsection (b) below, a successor shall be elected for a like term.
- (b) Within 30 days after the time that Retail Owners own 50% of the Baximum Units, or whenever the Class "3" Hember earlier determines, the Board shall be increased to five directors. The President shall call a special meeting at which Voting Members representing the Class "4" Members, other than the Declarant, shall be entitled to elect two of the five directors, who shall serve

as at-large directors. The remaining three directors shall be appointers of the Glass "5" Hember. The directors elected by the Voting Members shall not be subject to removal by the Class "8" Member and shall be elected for a term of two years or until the happening of the event described in subsection (c) below, whichever is shorter. If such directors' terms expire prior to the happening of the event described in subsection (c) below, successors shall be elected for a like term.

(c) Within 120 days after the termination of the Class "B" Control Period, the Board shall be increased to seven directors. The President shall call a meeting at which all directors shall be elected as follows: Subject to subsection (d) below, the directors shall be elected by Yoting Members representing both Class "A" and Class "B" Members. The majority of the directors shall be elected for a term of two years and the remaining directors shall be elected (or appointed as the case may be pursuant to subsection (d) below) for a term of one year, with each term to expire at the next annual meeting after the two-year or one-year period, as applicable. Successor directors shall be elected (or appointed as the case may be pursuant to subsection (d) below) at annual meetings to serve for two-year terms.

There shall be no cumulative voting. The candidates receiving the most votes shall be elected. For the first election held pursuant to this subsection, of the elected candidates, those receiving the most votes shall serve for two years. The directors elected by the Voting Members shall hold office until their respective successors have been elected. Directors may be elected to

- (d) After the Class "B" Control Period, the Declarant shall have the right to appoint or elect at least 20% of the Board Members as long as the Beclarant has the power to annex property pursuant to Article IX of the Declaration.
- 3.6 Removal of Birectors and Vacancies. Any director elected by the Voting Members may be removed, with or without cause, by the vote of Voting Members holding a majority of the votes entitled to be cast for the election of such director. Any director whose removal is sought shall be given notice prior to any meeting called and noticed for that purpose. A director who was elected solely by the votes of Voting Members representing Class "A" Members (other than the Beclarant), may be removed from Office prior to the expiration of his or her term only by the votes of Voting Members representing a majority of the Class "A" Members (other than the Beclarant). Upon removal of a director, a successor shall be elected by the Voting Members entitled to elect the director so removed to fill the vacancy for the remainder of the term of such director.

Any director elected by the Yoting Members who has three consecutive unexcased absences from Board meetings, or who is more than 30 days delinquent in the payment of any assessment or other charge due the Association, may be removed by a majority of the directors present at a regular or special meeting at which a quorom is present, and a successor may be appointed by the Board to fill the vacancy for the remainder of the term.

In the event of the death, disability, of resignation of a director, the Board may declare a vacancy and appoint a successor to fill the vacancy until the cent annual meeting, at which time the foling dembers entitled to fill such directorship may elect a successor for the remainder of the term.

8. Meetings

- 3.7 Organizational Meetings. The first meeting of the Board following each annual meeting of the membership shall be held within 10 days thereafter at such time and place as the Board shall fir.
- 3.8 Regular Meetings. Regular meetings of the Board may be held at such time and place as a majority of the directors shall determine, but at least four such meetings shall be held during each fiscal year with at least one per quarter. Notice of the time and place of the meeting shall be posted in a prominent place within the Properties and communicated to directors not less than four days prior to the meeting; provided, however, notice of a meeting need not be given to any director who has signed a waiver of notice or a written cossest to holding of the meeting.
- 3.9 Special Beetings. Special meetings of the Board shall be held when called by written notice signed by the Fresident or by any two directors. The notice shall specify the time and place of the meeting and the nature of any special business to be considered. The notice shall be given to each director by: (a) personal delivery; (b) first class mail, postage prepaid; (c) telephone communication, either directly to the director or to a person at the director's office or home who would reasonably be

expected to communicate such notice promptly to the director, or (d) telegram, charges prepaid. All such notices shall be given at the director's telephone number or sent to the director's address as shown on the records of the Association. Motices of special meetings of the Board shall be posted in a prominent place within the Properties. Motices sent by first class mail shall be deposited into a United States mailbox at least four business days before the time set for the meeting. Motices given by personal delivery, telephone, or telegraph shall be delivered, telephoned, or given to the telegraph company at least 72 bours before the time set for the meeting.

- 3.10 Maiver of Notice. The transactions of any meeting of the Board, however called and noticed or wherever held, shall be as valid as though taken at a meeting duly held after regular call and notice if (a) a quorum is present, and (b) either before or after the meeting each of the directors not present signs a written waiver of notice, a consent to holding the meeting, or an approval of the minutes. The maiver of notice or consent need not specify the purpose of the meeting. Motice of a meeting also shall be deemed given to any director who attends the meeting without protesting before or at its commencement about the lack of adequate notice.
- 3.11 Quorum of Board of Directors. At all meetings of the Board, a majority of the directors shall constitute a quorum for the transaction of business, and the votes of a majority of the directors present at a meeting at which a quorum is present shall constitute the decision of the Board, unless otherwise specifically provided in these By-Laws or the Declaration. A meeting at which a quorum is initially present may continue to transact business, notwithstanding the withdrawal of directors, if any action taken is approved by at least a majority of the required quorum for that meeting. If any meeting of the Board cannot be held because a quorum is not present, a majority of the directors present at such meeting may adjourn the meeting to a time not less than five nor more than 30 days from the date of the original meeting. At the reconvened meeting, if a quorum is present, any business which might have been transacted at the meeting originally called may be transacted without further notice.
- 3.12 <u>Compensation</u>. We director shall receive any compensation from the Association for acting as such unless approved by Voting Members representing a majority of the total Class "A" rotes in the Association, other than the Declarant, at a regular or special meeting of the Association. Any director may be reimbursed for expenses incurred on behalf of the Association upon approval of a majority of the other directors. Sothing herein shall prohibit the Association from compensating a director, or any entity with which a director is affiliated, for services or supplies furnished to the Association in a capacity other than as a director pursuant to a contract or agreement with the Association, provided that such director's interest was made known to the Board prior to entering into such contract and such contract was approved by a majority of the Board, excluding the interested director.
- 3.13 Conduct of Meetings. The President shall preside over all meetings of the Board, and the Secretary shall keep a minute book of meetings of the Board, recording all resolutions adopted by the Board and all transactions and proceedings occurring at such meetings.
- 3.14 Open Beetings. Subject to the provisions of Section 3.15, all meetings of the Board shall be open to all Voting Bembers, but a Voting Bember other than directors may not participate in any discussion or deliberation unless permission to speak is requested on his or her behalf by a director. In such case, the President may limit the time any Voting Bember may speak. Notwithstanding the above, the President may adjourn any meeting of the Board and recourses is executive session, excluding Voting Bembers, to discuss matters of a sensitive nature, such as pending or threatened litigation, personnel matters, etc.
- 3.15 Action Without a Formal Heeting. Any action to be taken at a meeting of the directors or any action that may be taken at a meeting of the directors may be taken without a meeting if a consent in writing, wetting forth the action so taken, shall be signed by all of the directors, and such consent shall have the same force and effect as a unanimous rote. Written consent or concents shall be filed with the minutes of the proceedings of the Board. An explanation of the action to be taken or actually taken by the Board shall be given to the Members of the Association within three days after all written consents have been obtained. The explanation shall be given in the same manner as provided in the By-Laws for the giving of notice of regular meetings of the Board. Failure to give notice shall not render the action to be taken or actually taken invalid.

C. Powers and Duties

3.16 Powers. The Board shall have all of the powers and duties necessary for the administration of the Association's affairs and for performing all responsibilities and exercising all rights of the Association as set forth in the Declaration, these By-Laws,

the Articles, and as provided by law. The Board may do or cause to be done all acts and things as are not by the Declaration, Articles, these By-Laws, or Oregon law directed to be done and exercised exclusively by the Poting Members or the membership generally.

- 3.17 Duties. The duties of the Board shall include without limitation;
 - (a) preparation and adoption of annual budgets and establishing each Owner's share of the Common Expenses;
 - (b) levying and collecting assessments from the Owners to fund the Common Expenses;
 - (c) providing for the operation, care, upkeep, and maintenance of the Area of Gommon Responsibility;
- (d) designating, hiring, and dismissing the personnel necessary to carry out the rights and responsibilities of the Association and where appropriate, providing for the compensation of such personnel and for the purchase of equipment, supplies, and materials to be used by such personnel in the performance of their duties;
- (e) depositing all funds received on behalf of the Association in a bank depository which it shall approve, and using such funds to operate the Association: provided, any reserve fund may be deposited, in the directors' best business judgment, in depositories other than banks;
 - (f) making and amending rules and regulations;
 - (g) opening of bank accounts on behalf of the association and designating the signatories required;
- (b) making, or contracting for the making of, repairs, additions, and improvements to, or alterations of, the Common Area in accordance with the Declaration and these By-Laws;
- (i) enforcing by legal means the provisions of the Declaration, these By-Laws, and the rules adopted by it and bringing any proceedings which may be instituted on behalf of, or against, the Owners concerning the Association;
- (j) obtaining and carrying property and liability insurance and fidelity bonds, as provided in the Declaration, paying the cost thereof, and filing and adjusting claims, as appropriate;
- (t) paying the cost of all service rendered to the Association or its Members and not chargeable directly to specific Owners;
 - (1) keeping books with detailed accounts of the receipts and expenditures of the Association;
- (m) making available to any prospective purchaser of a Unit, any Owner, and the holders, insurers, and guarantors of any Bortgage on any Unit, current copies of the Declaration, the Articles of Interporation, the By-Laws, rules and all other books, records, and financial statements of the Association;
- (n) permitting utility suppliers to use portions of the Common Area reasonably necessary to the ongoing development or operation of the properties;
- (o) indemnifying a director, officer or committee member, or former director, officer or committee member of the Association to the extent such indemnity is required by Oregon law, the Articles of Incorporation and the Duclaration; and
- (p) assisting in the resolution of disputes between owners and others without litigation, as set forth in the Beclaration.

3.18 Right of Class "B" Bember to Disapprove Actions. Until 75% of the maximum units are owned by Retail Owners, the Declarant shall have a right to disapprove any action, policy or program of the Association, the Board and any committee which, in the judgment of the Declarant, would tend to impair rights of the Declarant or Builders under the Declaration or these By-laws, or interfere with

development, construction of any portion of the Properties, or diminish the level of services being provided by the Association.

do such action, policy or program shall become effective or be implemented until and unless:

- (a) The Declarant shall have been given written notice of all neetings and proposed actions approved at meetings of the Association, the Board or any committee thereof by certified mail, return receipt requested, or by personal delivery at the address it has registered with the Secretary of the Association, as it may change from time to time, which notice complies as to the Board meetings with Sections 3.8, 3.9 and 3.10 of these By-Laws and which notice shall, except in the case of the regular meetings held pursoant to the Sy-Laws, set forth in reasonable particularity the agenda to be followed at said meeting; and
- (b) The Declarant shall be given the opportunity at any such meeting to join in or to have its representatives or agents join in discussion from the floor of any prospective action, policy, or program which would be subject to the right of disapproval set forth herein. The Declarant, its representatives or agents shall make its concerns, thoughts, and suggestions known to the Board and/or the members of the subject committee. The Declarant shall have and is hereby granted a right to disapprove any such action, policy, or program authorized by the Association, the Board or any committee thereof, if Board, committee, or Association approval is necessary for such action. This right may be exercised by the Declarant, its successors, assigns, representatives, or agents at any time within 10 days following the meeting held pursuant to the terms and provisions hereof. This right to disapprove may be used to block proposed actions but shall not extend to the requiring of any action or counteraction on behalf of any committee, or the Board or the Association. The Declarant shall not use its right to disapprove to reduce the level of services which the Association is obligated to provide or to prevent capital repairs or any expenditure required to comply with applicable laws and regulations.
- (c) This Section may not be amended without the express written consent of the Declarant until 75% of the Maximum Units have been conveyed to Retail Owners.
- 3.19 Management. The Board may employ for the Association a professional management agent or agents at such compensation as the Board may establish, to perform such duties and services as the Board shall authorize. The Board may delegate such powers as are necessary to perform the manager's assigned duties, but shall not delegate policy making authority or those duties set forth in Sections 3.17(a) and 3.17(i). The Beclarant, or no affiliate of the Declarant, may be employed as managing agent or manager.

The Board may delegate to one of its members the authority to act on behalf of the Board on all matters relating to the Duties of the managing agent or manager, if any, which might arise between meetings of the Board.

The Association shall not be bound, either directly or indirectly, by any management contract executed during the Class "B" Gontrol Period unless such contract contains a right of termination exercisable by the Association, with or without cause and without penalty, at any time after termination of the Glass "B" Control Period upon not more than 30 days written notice.

- 3.20 Accounts and Reports. The Following management standards of performance shall be followed unless the Board by resolution specifically determines otherwise:
 - (a) accornal accounting, as defined by generally accepted accounting principles, shall be employed;
 - (b) accounting and controls should conform to generally accepted accounting principles;
 - (c) cash accounts of the association shall not be commingled with any other accounts;
- (d) no remoneration shall be accepted by the managing agent from vendors, independent contractors, or others providing goods or services to the Association, whether in the form of commissions, finder's fees, service fees, prizes, gifts, or otherwise; anything of value received shall benefit the Association;
- (e) any financial or other interest which the managing agent may have in any firm providing goods or services to the Association shall be disclosed promptly to the Board;

- (f) the following financial and related information shall be regularly prepared and distributed by the Board to all Members of the Association:
- (i) The Board shall cause a "Capital Budget" and a "Common Sipense Budget" (collectively referred to as the "Budget") for the Association, to be prepared for each fiscal year of the Association, a copy of which shall be distributed to each Member of the Association not less than 45 nor more than 60 days before the beginning of the fiscal year to which the Sudget relates. A copy of the Budget shall be distributed personally or by mail or other manner reasonably designed to provide delivery to each of the Members of the Association. The Budget shall include the following information:
 - (A) the estimated revenue and expenses of the Association on an accrual basis for the forthcoming
- (5) the amount of the total cost receives of the Association currently available for the replacement or major repair of the Association Responsibility and for contingencies;

fiscal year;

- (C) an estimate of the current replacement costs of the estimated remaining useful life of, and the methods of funding to defray future repair, replacement or additions to, those major components of the Area of Common Responsibility; and
- (D) a general statement setting forth the procedures used by the Board in the calculation and establishment of reserves to defray the future repair, replacement or additions to major components of the Area of Common Responsibility.
- (B) a disclosure statement that the Declarant or a Builder is contributing "in kind" services or material pursuant to a contract with the Association, if any, and that their assessment obligation may be reduced or materials; and
- (7) a disclosure statement that a subsidy contract exists between the Declarant and the Association, if that is the case.

The Budget shall become effective unless disapproved at a meeting of the Voting Members representing at least a majority of the Members of the Association. There shall be no obligation to call a meeting for the purpose of considering the Budget except on petition of the Voting Members as provided for special meetings in Section 2.4 of the By-Laws, which petition must be presented to the Board within 10 days of delivery of the notice of assessments. In the event the proposed budget is disapproved or the Board fails for any reason to determine the Budget for any year, then and until such time as a Budget shall have been determined as provided herein, the Budget in effect for the immediately precenting year shall continue for the current year.

In lieu of distributing the Budget as specified in the above paragraphs of this Section 3.20, the Board may elect to distribute a summary of the Budget ("Summary") to all its Members with a written notice that the Budget is available at the business office of the Association or at another suitable location within the boundaries of the Properties and that copies will be provided upon request at the expense of the Association. If any Member requests a copy of the Budget, the Association shall provide such copy to the Member by first-class United States mail and deliver such copy within five days of such request. The written notice that is distributed to each of the Members shall be in at least 10 point hold type on the front page of the Summary.

In the event the Board elects to distribute the Budget, upon written request for the Association's most recent Budget by any Member, the Association shall, within 10 days of the mailing or delivery of the request, provide to any Member of the Association a copy of the Budget. The Association may charge a fee for this service, which shall not exceed the reasonable cost to prepare and reproduce the Budget.

(ii) The Board shall cause an amnual report ("Financial Statement") to be prepared in accordance with generally accepted accounting principles to be distributed to each Member of the Association within 120 days after close of the Association's fiscal year. A copy of the Financial Statement shall be distributed personally or by mail or other manner reasonably designed to provide delivery to each of the Members of the Association, and shall consist of:

- (4) a balance sheet as of the end of the fiscal year;
- (B) an income and expense statement for the fiscal year (This operation Statement shall include as schedule of assessments received and receivables identified by the numbers of the Units and the names of the Owners assessed.);
 - (C) a statement of changes in financial position for the fiscal year.

The Pinancial Statement shall be prepared by an independent accountant for any fiscal year in which the gross income to the Association exceeds \$75,000.00. If the Pinancial Statement is not prepared by an independent accountant, it shall be accompanied by the certificate of an authorized officer of the Association that the statements were prepared without independent audit or review of the books and records of the Association.

- (iii) The Board shall do the following not less frequently them quarterly:
 - (A) cause a current reconciliation of the Association's operating accounts to be made and review

the same;

- (B) cause a current reconciliation of the Association's reserve accounts to be made and review the same;
 - (C) review the current year's actual reserve revenues and expenses compared to the current year's

Budget;

and

- (D) review the most current account statements prepared by the financial institution where the Association bas its operating and reserve accounts;
 - (8) review an income and expense Statement for the Association's operating and reserve accounts;

(F) review the deliaquency report listing all Owners who are deliaquent in paying any assessments at the time of the report and describing the status of any action to collect such assessments which remain deliaquent. (Any assessment or installment thereof shall be considered to be deliaquent on the 15th day following the due date unless otherwise determined by

- (iv) A statement of the Association's policies and practices in enforcing its remedies against Members for default in the payment of assessments, including the recording and foreclosing of liens against Units, to be distributed to Members of the Association within 60 days before the beginning of each fiscal jear.
- 3.21 Borrowing. The Association shall have the power to borrow money for the purpose of maintenance, repair or restoration of the Area of Common Responsibility without the approval of the Voting Members. The Board shall also have the power to borrow money for other purposes; provided, that the proposed borrowing is for the purpose of modifying, improving or adding amenities to the Area of Common Responsibility. Such borrowing shall be in accordance with Section 3.24(a). During the Class "B" Control Period, no Mortgage lien shall be placed on any portion of the Common Area without the affirmative vote or written consent of Voting Hembers representing at least a majority of the Members other than the Declarant.
- 3.22 Rights of the Association. The Association shall have the right to contract with any Person for the performance of various duties and functions. This right shall include, without limitation, the right to enter into common management, operational, or other agreements with trusts, condominiums, cooperatives, other owners or residents associations, both within and outside the Properties. Such agreements shall require the consent of a majority of the total number of directors of the Association.
 - 3.23 <u>Referencent.</u> In addition to such other rights as are specifically greated under the Decisration, the Board shall have

the power to impose reasonable fines. A fine, other than a penalty for nonpayment of assessments, shall not constitute a lien upon the Unit of the violating Owner. The Board shall also have the power to suspend an Owner's right to rote or any person's right to use the Common Area for violation of any duty imposed under the Declaration, these By-Laws, or any rules and regulations duly adopted hereunder; provided, however, nothing herein shall authorize the Board to limit regress and agrees to or from a Unit or to suspend an Owner's right to rote due to nonpayment of assessments. In addition, the Board may suspend any service provided by the Association to an Owner or the Owner's Unit if the Owner is more than 30 days delinquent in paying any assessment or other charges owed to the Association. In the event that any occupant, guest or invitee of a Unit violates the Declaration, By-Laws, or a rule and a fine is imposed, the fine shall first be assessed against the occupant; provided, however, if the fine is not paid by the occupant within the time period set by the Board, the Owner shall pay the fine upon notice from the Association. The failure of the Board to enforce any provision of the Beclaration, By-Laws, or any rule shall not be deemed a waiver of the right of the Board to do so thereafter.

- (a) <u>Notice</u>. Prior to imposition of any manction hereunder or under the Declaration, the Board or its delegate shall serve the alleged violator with written notice describing (i) the nature of the alleged violation, (ii) the proposed sanction to be imposed, (iii) a period of not less than 15 days within which the alleged violator may present a written request for a hearing to the Board or the Covenants Committee, if any, appointed pursuant to Article 7; and (iv) a statement that the proposed sanction shall be imposed as contained in the notice noless a challenge is begun within 15 days of the notice. If a timely challenge is not made, the sanction stated in the notice shall be imposed; provided the Board or the Covenants Committee may, but shall not be obligated to, suspend any proposed sanction if the violation is cured within the 15 day period. Such suspension shall not constitute a waiver of the right to sanction future violations of the same or other provisions and rules by any Person.
- (b) Staring. If a hearing is requested within the allotted 15 day period, the hearing shall be held before the Covenants Committee, if any, or if none, before the Board in executive session. The alleged violator shall be afforded a reasonable opportunity to be heard. Prior to the effectiveness of any sanction bereunder, proof of proper notice shall be placed in the minutes of the meeting. Such proof shall be deemed adequate if a copy of the notice, together with a statement of the date and manner of delivery, is entered by the officer, director, or agent who delivered such notice. The notice requirement shall be deemed satisfied if the alleged violator appears at the meeting. The minutes of the neeting shall contain a written statement of the results of the hearing and the sanction, if any, imposed.
- (c) <u>Appeal.</u> Pollowing a hearing before the Covenants Committee, the violator shall have the right to appeal the decision to the Board. To perfect this right, a written notice of appeal must be received by the manager, President, or Secretary of the Association within 10 days after the hearing date.
- (d) Additional Enforcement Rights. Notwithstanding anything to the contrary in this Article, the Board may elect to enforce any provision of the Declaration, these By-Laws, or the rules of the Association by self-help (specifically including, but not limited to, the towing of vehicles that are in violation of parking rules and regulations; however, only in accordance with any applicable ordinances of the County of Deschutes, if applicable) or, following compliance with the procedures set forth in Article XYI of the Declaration, by suit at law or in equity to enjoin any violation to recover monetary damages or both without the necessity of compliance with the procedure set forth above. In any such action, to the maximum extent permissible, the Owner or occupant responsible for the violation of which abatement is sought shall pay all costs, including reasonable attorney's fees actually incurred.
- 3.24 <u>Prohibited Acts.</u> The Board shall not take any of the following actions except with the written consent or vote of the Yoting Members representing at least a majority of the Benbers other than the Declarant:
- (x) to incur aggregate expenditures for capital improvements to the Area of Common Responsibility in any fiscal year in excess of 51 of the budgeted Common Expenses of the Association for that Piscal year;
- (b) to sell during any fiscal year property of the Association baving an aggregate fair market value greater than 5% of the budgeted Common Expenses of the Association for that fiscal year;
- (c) to pay compensation to directors or officers of the Association for service performed in the conduct of the Association's business; provided, however, the Board may cause a director or officer to be reimbursed for expenses incurred in carrying on the business of the Association;

- (d) to enter into a contract with a third Person wherein the third Person will turnish goods or services for a term longer than one year with the following exceptions:
- (i) a management contract, the terms of which have been approved by the U. S. Department of Veterans Affairs the Properties:
- (ii) a contract with a public utility company, if the rates charged for the materials or services are regulated by the Public Utilities Commission; provided, however, the terms of the contract shall not exceed the shortest term for which a supplier will contract at the regulated rate;
- (iii) prepaid casualty and/or liability insurance policies not to exceed three years duration, provided that the policy permits short rate cancellation by the insured;
- (iv) lease agreements for laundry room fixtures and equipment of a duration not to exceed five years, provided that the lessor is not an entity in which Declarant has a direct or indirect interest of 10% or more;
- (v) agreements for cable television services and equipment or satellite television services and equipment of the exceed five years duration, provided that this supplier is not an entity in which Declarant has a direct or indirect interest of 10% or more;
- (vi) agreements for male or lease of burglar alarm and fire alarm equipment, installation and services thereof, not to exceed five years duration, provided that the supplier is not an entity in which Declarant has a direct or indirect interest of 10% or more.
- (e) no contract with the Association negotiated by Declarant shall exceed a term of one year except as may otherwise be provided in paragraph (d) of this Section.

ARTICLE IV Officers

- 4.1 Officers. The officers of the Association shall be a President, Vice President, Secretary, and Treasurer. The President, Vice President, Secretary and Treasurer shall be elected from among the members of the Board. The Board may appoint such other officers, including one or more Assistant Secretaries and one or more Assistant Treasurers, as it shall deep desirable, such officers to have the authority and perform the duties prescribed by the Board. Any two or more offices may be held by the same person, except the offices of President and Secretary.
- 4.2 <u>Election and Term of Office</u>. The officers of the Association shall be elected annually by the Board at the first meeting of the Board following each annual meeting of the membership, as set forth in Article III.
- 4.3 Removal and Vacancies. Any officer may be removed by the Board whenever in its judgment the best interests of the Association will be served thereby. A Vacancy in any office arising because of death, resignation, removal, or otherwise may be filled by the Board for the unexpired portion of the term.
- 4.4 Fowers and Duties. The officers of the Association shall each have such powers and duties as generally pertain to their respective offices, as well as such powers and duties as may specifically be conferred or imposed by the board. The President shall be the chief executive officer of the Association. The Treasurer shall have primary responsibility for the preparation of the budget as provided for in the Declaration and may delegate all or part of the preparation and notification duties to a finance committee, management agent, or both.
- 4.5 <u>Resignation</u>. Any officer may resign at any time by giving written notice to the Board, the President, or the Secretary. Such resignation shall take effect on the date of the receipt of such motive or at any later time specified therein, and unless

otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.

- 4.6 Agreements, Contracts, Beeds, Leases, Cheeks, Brc. All agreements, contracts, deeds, leases, cheeks, and other instruments of the Association (other than for the withdrawal of reserve funds) shall be executed by at least two officers or by such other person or persons as may be designated by resolution of the Board. The Board shall require signatures for the withdrawal of reserve funds of either two members of the Board or a member of the Board and officer of the Association who is not also a member of the Board. Por purposes of this section, "reserve funds" means monies the Board has identified in the Budget to defray the future repair or replacement of, or additions to, those major items which the Association is obligated to maintain.
- 4.7 Compensation. Compensation of officers shall be subject to the same limitations as compensation of directors under Section 3.12 hereof.

ARTICLE V Committees

- 5.1 General. The Board may appoint such committees as it deems appropriate to perform such tasks and to serve for such periods as the Board may designate by resolution. Bach committee shall operate in accordance with the terms of such resolution.
- 5.2 Covenants Committee. In addition to any other committees which the Board may establish pursuant to Section 5.1, the Board may appoint a Covenants Committee consisting of at least three and no more than seven Bembers. Acting in accordance with the provisions of the Declaration, these By-Laws, and resolutions the Board may adopt, the Covenants Committee, if established, shall be the hearing tribunal of the Association and shall conduct all hearings held pursuant to Section 3.23 of these By-Laws.
- 5.3 Transitional Advisory Committee. In addition to any other committees appointed as provided above, the Board shall establish persuant to 0.8.8. 96.604, a Transitional Advisory Committee when more than SOZ of the Hazimum Units are owned by Retail Owners and Class "B" membership has not yet terminated.

ARTICLE VI Biscellaneous

- 6.1 <u>Fiscal Year.</u> The fiscal year of the Association shall be set by resolution of the Board. In the absence of a resolution, the fiscal year shall be the calendar year.
- 6.2 Parliamentary Rules. Except as may be modified by Board resolution, Robert's Rules of Order (current edition) shall govern the conduct of Association proceedings when not in conflict with Oregon laws, the Articles of Incorporation, the Declaration, or these By-Laws.
- 6.3 Conflicts. If there are conflicts between the provisions of Oregon law, the Articles of Incorporation, the Declaration, and these By-Laws, the provisions of Oregon law, the Declaration, the Articles of Incorporation, and the By-Laws (in that order) shall prevail.

6.4 Books and Records.

(a) <u>laspection by Members and Mortgagees</u>. The Board shall make available for inspection and copying by any holder, insurer or guarantor of a first Mortgage on a Unit, any Member, or the duly appointed representative of any of the foregoing at any reasonable time and for a purpose reasonably related to his or her interest in a Unit: the Declaration, By-Laws, and Articles of Incorporation, any amendments to the foregoing, the rules of the Association, the membership register, the most recent Financial Statement, the current Budget, books of account, and the minutes of meetings of the Members, the Board, and committees. The Board shall provide for such inspection to take place at the office of the Association or at such other place within the Properties as the Board shall designate.

- (b) Rules for Inspection. The Board shall establish reasonable rules with respect to:
 - (i) notice to be given to the costodian of the records;
 - (ii) hours and days of the week when such an inspection may be made; and
 - (iii) payment of the cost of reproducing copies of documents requested.
- (c) <u>Inspection by Directors</u>. Brery director shall have the absolute right at any reasonable time to inspect all books, records, and documents of the Association and the physical properties owned or controlled by the Association. The right of inspection by a director includes the right to make a copy of relevant documents at the expense of the Association.
- 6.5 <u>dotices.</u> Unless otherwise provided in these By-Lams, all notices, demands, bills, statements, or other communications under these By-Laws shall be in writing and shall be deemed to have been duly given if delivered personally or if sent by United States mail, first class postage prepaid;
- (a) if to a Member or Voting Member, at the address which the Member or Voting Member has designated in writing and filed with the Secretary or, if no such address has been designated, at the address of the Unit of such Member or Toting Member; or
- (b) if to the Association, the Board or the managing agent, at the principal office of the Association or the managing agent, if any, or at such other address as shall be designated by notice in writing to the Members pursuant to this Section.

6.6 Amendment.

- (a) By Declarant Member. Prior to the conveyance of the first Unit to a Retail Owner and prior to Commencement of Sales, the Beclarant may unilaterally amend these By-Laws. After the conveyance of any Doit, the Declarant may unilaterally amend these By-Laws at any time if such amendment is;
 - (i) mecessary to bring any provision hereof into compliance with any applicable governmental statutes, rule or regulation, or judicial determination;

(ii) necessary to enable any reputable title insurance company to issue title insurance coverage on the Voits:

(iii) required by an institutional or governmental lender or purchaser of mortgage loans, including, for example, the federal Mational Mortgage Association or Federal Nome Loan Mortgage Corporation, to enable such lender or purchaser to make or purchase mortgage loans on the Units; or

(ir) otherwise necessary to enable any governmental agency or reputable private insurance company to insure mortgage loans on the Units; provided, however, any such amendment shall not adversely affect the title to any Unit unless the affected Owner shall consent thereto in writing.

(b) By Humbers Generally. Except as provided above, these By-Laws may be amended only by the affirmative vote or written consent, or any combination thereof, of Toting Members representing SLI of each class of Members. After conversion of the Class "B" membership to Class "A" membership, these By-Laws may be amended by the vote or written consent of the Voting Members representing at least: (a) 75% of the Members; and (b) 75% of the Members other than the Declarant.

In addition, the approval requirements set forth in article XIV of the Declaration shall be met, if applicable. Hotwithstanding the above, the percentage of votes necessary to amend a specific clause shall not be less than the prescribed percentage of affirmative votes required for action to be taken under that clause.

(c) Validity and Effective Date of Amendments. Amendments to these By-Laws shall become effective upon recordation in the Office of The County Clerk galess a later effective date is specified therein. Any procedural challenge to an amendment must be made within six months of its recordation or such amendment shall be presumed to have been validly adopted. In no event shall a change of conditions or circumstances operate to amend any provisions of these By-Laws.

If an Owner consents to any amendment to the Declaration or these By-Laws, it will be conclusively presumed that such Owner has the Authority so to consent and no contrary provision in any Mortgage or contract between the Owner and a third party will affect the validity of such amendment.

Bo amendment may remove, revoke, or modify any right or privilege of Declarant without the written consent of Declarant or the assignee of such right or privilege.

CERTIFICATION

	I, the undersigned, do hereby certify:
	That I am the duly elected and acting Secretary of the Association;
held	That the foregoing By-Laws constitute the original By-Laws of said Association, as duly adopted at a meeting of the Board thereof on the
	IN WITHESS WHEREOF, I have becomen subscribed my name and affixed the seal of said Association thin day
	Secretary

EXHIBIT "D"

Rules of Arbitratico

- Claiment shall submit a Claim to arbitration under these Roles by giving written notice to all other Parties stating plainly
 and concisely the nature of the Claim, the remedy sought and Claimant's desire to submit the Claim to arbitration ("Arbitration").
- 2. Bath Party shall select so arbitrator ("Party Appointed Arbitrator"). The Party appointed Arbitrators shall, by agreement, select one or two neutral arbitrators ("Mentral(s)") so that the total arbitration panel ("Panel") has an odd number of arbitrators. If any Party fails to appoint Party Appointed Arbitrator within 20 days from the date of the Arbitration Notice, the remaining arbitrators shall conduct the proceedings, selecting a Neutral in place of any missing Party Appointed Arbitrator. The Seutral arbitrator(s) shall select a chairperson ("Chair").
- 3. If the Panel is not selected under Rule 2 within 4) days from the date of the Arbitration Notice, Claimant may notify the Oregon chapter of The Community Association Institute, which shall appoint one Meutral ("Appointed Meutral"), notifying the Appointed Neutral and all Parties in writing of such appointment. The Appointed Neutral shall thereafter be the sole arbitrator ("Arbitrator"), and any Party Appointed Arbitrators or their designees shall have no further duties involving the arbitration proceedings.
- 4. Ho person may serve as a Neutral in any arbitration under these Rules in which that person has any financial or personal interest is the result of the arbitration. Any person designated as a Neutral shall immediately disclose in writing to all Parties any circumstances likely to affect impartiality, including any bias or financial or personal interest in the outcome of the arbitration shall be replaced in the same manner in which that Neutral was selected.
- 5. The Arbitrator or Chair, as the case may be, ("Arbitrator") shall fix the date, time, and place for the hearing. The place of the hearing shall be within the Properties unless otherwise agreed by the Parties.
 - Any Party may be represented by an attorney or other authorized representative throughout the arbitration proceedings.
- 7. All persons who, in the judgement of the Arbitrator, have a direct interest in the arbitration are entitled to attend hearings.
 - 8. There shall be no stemographic record of the proceedings.
- 9. The hearing shall be conducted in whatever manner will, in the Arbitrator's judgement, most fairly and expeditionally permit the full presentation of the evidence and arguments of the Parties.
- 10. The Parties may offer such evidence and material as is relevant to the Claim, and shall produce such additional evidence as the Arbitrator may deem necessary to reach an understanding and determination of the Claim. The Arbitrator shall be the sole judge of the relevance and materiality of any evidence offered, and conformity to the legal rules of evidence shall not be necessary. The Arbitrator shall be authorized, but not required, to administer oaths to witnesses.
 - 11. The Arbitrator shall declare the bearings closed when satisfied the record is complete.
 - 12. There will be no posthearing briefs.

Commence of the second second

- 13. The Award shall be rendered immediately following the close of the hearing, if possible, and no later than 14 days from the close of the hearing, unless otherwise agreed by the Parties. The Award shall be in writing, shall be signed by the Arbitrator and acknowledged before a notary public. If the Arbitrator believes an opinion is necessary, it shall be in summary form.
 - 14. If there is more than one arbitrator, all decisions of the Panel and the Award shall be by majority vote.
- 15. Each Party agrees to accept as legal delivery of the Award the deposit of a true copy in the mail addressed to that Party or its attorney at the address communicated to the Arbitrator at the hearing.

SUPPLEMENTAL DECLARATION TO ANNEX CERTAIN PROPERTY AND SUBJECT THE SAME TO CERTAIN COVENANTS

Pursuant to Sections 1.17 and 9.1 of the amended Covenants and Restrictions of the Golf Course Estates at Aspen Lakes Subdivision which were recorded in Book 292, Pages 713-769 on March 9, 1993, in the Deschutes County Records, the Declarant hereby subjects the property described below to the original covenants and restrictions referred to above as they may have been amended from time to time:

Golf Course Estates at Aspen Lakes Phase IV, Deschutes County, Oregon, as shown on the official plat thereof as described as follows: Lots Ninety-Eight (98) through One Hundred Seventeen (117), Golf Course Estates at Aspen Lakes Phase IV, Deschutes County, Oregon.

DATED: Satember == 2, 2006.

MATTHEW CYRUS, Managing Member Aspen Lakes, LLC, Declarant

STATE OF OREGON

County of Deschutes

On <u>Sept. 22</u>, 2006, personally appeared before me the above named

Matthew Cyrus, Managing Member of Aspen Lakes, LLC, Declarant, and acknowledged the foregoing instrument to be his voluntary act.

OFFICIAL SEAL
SHELLEY MARSH
NOTARY PUBLIC-OREGON
COMMISSION NO. A370783
MY COMMISSION EXPIRES JULY 22, 2007

SMULL Marsh Notary Public for Oregon

DESCHUTES COUNTY OFFICIAL RECORDS NANCY BLANKENSHIP, COUNTY CLERK

2006-64522

00496710200600645220010011

\$31.00

D-CCR Cnt=1 Stn=1 BN \$5.00 \$11.00 \$10.00 \$5.00

After recording, return to
Amerititle
15 OREGON AVENUE, BEND

Supplemental Declaration

MERRILL O'SULLIVAN, LLP

ATTORNEYS AT LAW 1070 NW BOND STREET, SUITE 303 BEND, OR 97701

Page 1

09/22/2006 04:04:17 PM

\\Server\data\\WPDOCS\MMM\CLIENTS\Cyrus\Sup Dec 9-1-06.doc